# Does Political Influence Distort Banking Regulation? Evidence from the US

Panagiota Papadimitri<sup>a</sup>, Fotios Pasiouras<sup>b</sup>, Gioia Pescetto<sup>a</sup>, and Ansgar Wohlschlegel<sup>c</sup>

> <sup>a</sup>Portsmouth Business School, University of Portsmouth, Portsmouth, UK <sup>b</sup>Montpellier Business School, Montpellier, France <sup>c</sup>University of Swansea, UK

This study examines the interplay between political influence and regulatory decisionmaking.Political influence is captured based on whether a bank is headquartered in a state where an elected official holds a chair position on a congressional committee related to the banking and financial services industry. Using data of US commercial banks over the period 2000-2015, we show that our measure of political influence reduces a bank's probability of receiving a formal regulatory enforcement action. Results are robust to the use of alternative model specifications and the sample restrictions. However, we find that various bank and environmental characteristics are important conditional factors.

**Keywords:** Political Influence · Congressional Committees · Banking Supervision · Enforcement Actions

<sup>\*</sup> We would like to thank two anonymous reviewers and Iftekhar Hasan (Editor) for valuable comments that helped us improve the manuscript. Special thanks are also due to Barbara Casu (conference discussant), Patrycja Klusak (conference discussant), Chris Tsoumas and all the participants at the 8th International Conference of the Financial Engineering and Banking Society, 5th Young Finance Scholars Conference, EFiC Conference in Banking and Finance 2018, Portsmouth Business School Research Seminar Series, Portsmouth Business School PGRS conference, the 1st International Symposium in Finance, the International Hellenic University's Research Seminar Series, and the 7th International Conference on Multidimensional Finance, Insurance and Investment. Montpellier Business School (MBS) is a founding member of the public research center Montpellier Research in Management, MRM (EA 4557, Univ. Montpellier). Any remaining errors are our own.

<sup>&</sup>lt;sup>‡</sup> Contact author. Mailing Address: Montpellier Business School, Montpellier, France. E-mail: f.pasiouras@montpellier-bs.com

E-mail(s): panagiota.papadimitri@myport.ac.uk, gioia.pescetto@port.ac.uk, a.u.m.wohlschlegel@swansea.ac.uk

"[...] the connection between politics and finance is complex. It differs across countries with different political and financial systems. It changes over time within countries, as some constituencies gain power, others lose power, and political and financial systems evolve."

- Barth *et al.* (2012, p.17)

## 1 Introduction

The regulatory and supervisory architecture of the banking sector has received extensive attention, with interest therein being renewed during the crisis and the post-crisis period. While policy makers and academics have traditionally focused on prudential - and most recently macroprudential - regulations, attention has, since the financial crisis, increasingly been shifted to the role of political influence, which runs high on the agenda of the banking industry.

The interest in this latter issue is not surprising for at least two reasons. First, regulatory capture and political influence has been cited as one of the reasons that led to the crisis. For instance, the 2011 report of the national commission on the causes of the financial and economic crisis in the US concludes that "widespread failures in financial regulation and supervision proved devastating to the stability of the nation's financial markets. More than 30 years of deregulation and reliance on self-regulation by financial institutions, championed by former Federal Reserve chairman Alan Greenspan and others, supported by successive administrations and Congress, and actively pushed by the powerful financial industry at every turn, had stripped away key safeguards, which could have helped avoid catastrophe" (p. xviii). Second, the high level of regulatory oversight and policy complexities in the banking industry imply that political connections, lobbying, and influence, may be particularly valuable (Gropper et al., 2015; Igan and Lambert, 2018). In fact, according to data provided by the Center for Responsive Politics, the financial sector ranks amongst the top spenders in

lobbying and campaign contributions. Igan and Lambert (2018) also point out that between 1998 and 2016 the U.S. financial sector spent \$7.4 billion on lobbying. Most importantly, during the 2016 US election cycle, the dollar amount of lobbying and campaign contributions by large banks, hedge funds and other financial institutions hit a record high approximately 25% higher than the amount spent during the 2007-2008 elections (McLannahan and Jopson, 2017)

Focusing on the US banking sector, the main objective of the present study is to examine whether and how political influence impacts the decision making of financial regulators. As in Lambert (2019) we consider regulatory enforcement actions issued against banks as a proxy of regulatory decisions. In his study, which is the most closely related one to our work, he finds that lobbying banks are less likely to receive regulatory enforcement actions. However, we differentiate our work from the one of Lambert in some crucial respects. The most important one is that, while he focuses on bank lobbying activities, we investigate the impact of the political influence of elected officials on regulatory decisions. This is particularly important for at least two reasons. First, Igan and Lambert (2018) highlight that, while the size of the bank lobbying market appears to be big, the expenditures are actually rather small when compared to the size of the banking industry, the value of policies, and the benefits at stake. One potential explanation that they provide is that special interest groups do not need to spend much once they have reached out to the legislator or regulator and provided some key information. As an example, they mention that for a congressman, the relevant piece of information could be the impact of his vote for or against a particular bill on his re-election prospects. Another explanation that they offer is that lobbying expenditures are supplemented by other mechanisms of influence like quid pro quo agreements, career concerns, relationships, and persuasion. In the present study, to account for such factors, political influence is captured by whether or not a bank's headquarters are located in a state represented by a Senator or Congressman who is a Chairman in a Committee relevant to banking and financial services. The second reason for which the impact of the political

influence of elected officials on regulatory and governmental decisions is important is that it has received considerably less attention compared to firms' corporate political strategies, such as lobbying (e.g. Lambert, 2019), campaign contributions (e.g. Correia, 2014) or employment connections (e.g. Shive and Forster, 2016).

Therefore, we attempt to shed more light on one of the mechanisms through which political influence may work. This approach makes our work related to recent studies by Gropper et al. (2013, 2015) who use the same proxy for political influence to conclude that it "pays" to have friends in high places. However, these two studies focus on bank profitability and stock market performance and not on regulatory decisions. In contrast, by focusing on regulatory enforcement actions against banks headquartered in an elected official's home state, the present study provides direct evidence on a channel via which this elected official may exert the political influence associated with holding a chair position in a congressional committee (House or Senate) relevant to the financial industry. Shielding banks from regulatory enforcement actions is a plausible channel for a committee chair to exert his influence for two reasons: First, enforcement actions must be appropriately "tailored" around the unique features and needs of each institution. Essentially, this means that the regulators have some flexibility on the severity and type of action that will be imposed. Second, the regulators have in many instances worked in the past with members of the Senate or they have been nominated and supported by them <sup>1</sup>.

<sup>&</sup>lt;sup>1</sup>Take for example, Martin J. Gruenberg, the 20th Chairman of the FDIC, who received Senate confirmation on November 15, 2012. Previously he held other important positions in the FDIC board, and before that he had broad congressional experience in the financial services and regulatory areas. He served as Senior Counsel to Senator Paul S. Sarbanes (D-MD) on the staff of the Senate Committee on Banking, Housing, and Urban Affairs from 1993 to 2005. Mr. Gruenberg advised the Senator on issues of domestic and international financial regulation, monetary policy and trade. He also served as Staff Director of the Banking Committee's Subcommittee on International Finance and Monetary Policy from 1987 to 1992. As another example, on July 27 2017, U.S. Senator Mike Crapo (R-Idaho), Chairman of the United States Senate Committee on Banking, Housing and Urban Affairs, delivered the following remarks during a full committee hearing on the nominations of Mr. Joseph Otting, of Nevada, to be Comptroller of the Currency; and The Honorable Randal Quarles, of Colorado, to be a Member of the Board of Governors of the Federal Reserve System."Mr. Otting brings a particular expertise and understanding of our banking system from a long career in financial services. I am confident that Mr. Otting will bring strong leadership to the OCC" and "Mr. Quarles has a wealth of government and private sector experience dealing with both domestic and international financial markets. He is no stranger to public service, having previously served in multiple top posts in the Treasury Department".

To fulfill the objective of the study, we hand-collect data for US Commercial Banks and biographical information of Congressional Committee Chairmen from the period 2000-2015. Overall, the results confirm that banks with political connections are less likely to receive a severe type regulatory enforcement action. These results hold when we perform a set of estimations to examine the robustness of our results. First, we apply propensity-score matching in order to establish causality and rule out potential selection bias in the setting of our analysis. Second, we include alternative control variables and examine whether the results hold for several sub-samples (e.g. crisis vs non-crisis years, large banks, poor vs top capitalized banks). In all these cases, the results remain the same. Third, we examine alternative factors that could enhance our understanding of political channels or influence the reported results. Briefly, our findings suggest that in cases where the Chair's seniority is higher, the financial sector's contributions are higher, and the level of vote concentration is greater, the impact of local political influence on regulatory enforcement likelihood is enhanced. Furthermore, our findings suggest that although committee membership exerts a significant impact on regulatory enforcement likelihood; the impact of sub-committee chairmanship is only marginally significant. In terms of bank-related traits, the results indicate that the committee chair dummy is negative and statistically significant for both listed and non-listed parent companies. When we examine parent company age, we find that our core variable of interest is highly significant in the case of older than average parent companies, whereas the chair effect is weaker for younger parent companies. Moreover, we find that political influence is statistically significant only in the case that operate in a single State. Last but not least, we find that the overall state environment plays a role in our empirical investigation. In particular, political influence appears to matter particularly in states with higher corruption, lower levels of religiosity, and higher polarization.

Empirical precedent has demonstrated the linkage between politics and various aspects of today's corporate and banking sector environment. Empirical and theoretical studies overall agree that political connections are a "valuable" investment (Mathur et al., 2013), in the sense that firms with identified connections are more likely to enjoy certain benefits. In particular, the literature has indicated beneficial outcomes such as easier access to finance (Infante and Piazza, 2014; Chen et al., 2014), as well as lower cost of bank loans (Houston et al., 2014). In addition, studies provide evidence supporting that political connections lead to increased performance (Johnson and Mitton, 2003; Gropper et al., 2013, 2015; Papadimitri et al., 2019b) and corporate value (Faccio et al., 2006; Cooper et al., 2010; Hill et al., 2013). Moreover, studies have also explored other areas of corporate activity and find that political connections is tied to lower cost of capital (Boubakri et al., 2012) and milder depositor reaction (Disli et al., 2013; Nys et al., 2015). On a more general level, empirical precedent has also demonstrated that political influence is connected with higher stock market performance (Coates and Wilson, 2007) and economic growth (Coates et al., 2011).Finally, Duso (2005), who examine the interplay between lobbying and regulation, provide evidence suggesting that successful lobbying leads to avoidance of regulation in cases where otherwise it would have been compelling. Interestingly enough, their findings also suggest that a market's regulatory regime is tied both to the political and regulatory environment in place.

Our research is related to studies examining the interplay between political economy and general regulation. This literature dates back to Stigler (1971),Posner (1974), and Peltzman (1976), the pioneers of the theoretical models of regulatory influence. Besley and Coate (2001) and Grossman and Helpman (2001) provide theoretical models on how firms can influence policy outcomes. Other studies (e.g. Mailath and Mester, 1994; Acharya and Yorulmazer, 2007) have explored why and to what extent regulators use their discretion. More recently, Holburn and Bergh (2008) develop a theoretical argument suggesting that firms operating in heavily-regulated industries, have a higher probability to engage in corporate political strategies, in order to influence changes in legislation/laws towards their own interest.Gibson et al. (2018) and Kerr et al. (2014) examine the determinants and dynamics of lobbying activities, respectively.

In addition, our work is related to a growing number of empirical studies providing evi-

dence that regulatory decisions and firm outcomes are subject to political influence exerted by organizations. Many of them focus on SEC enforcement actions (Kedia and Rajgopal, 2011; Correia, 2014; Fulmer et al., 2017) while others examine fraud detection and enforcement against fraud (Yu and Yu, 2011; Wu et al., 2016), tax enforcement effectiveness (Lin et al., 2018), public reports of major violations for safety compliance (Fisman and Wang, 2015), government procurement contracts (Goldman et al., 2013), corporate bailouts (Faccio et al., 2006), preferential access to finance (Claessens et al., 2008) and Federal Reserve lending (Blau, 2017). However, only a limited number of studies focus on the banking industry. In an early study, Kroszner and Strahan (1999) examine bank branching deregulation in 1970s and 1980s in US and find that pressure exerted by special interest groups had an impact on the elimination of restrictions on bank branching. Along the same lines, Igan and Mishra (2014) suggest that higher lobbying from financial sector encouraged deregulation during the pre-crisis period. Focusing on the financial crisis, Mian et al. (2010) find that higher financial sector contributions increased the probability of Congressional members supporting bailout legislation of 2008. In a somehow related study, Duchin and Sosyura (2012) find that politically connected financial firms are more likely to be funded through the Capital Purchase Program (CPP). Examining the lending pattern of lobbying and non-lobbying firms, Igan et al. (2012) reveal that institutions that lobby on issues related to mortgage lending and securitization engaged in more risky lending practices ex ante and had worse performance ex post. Somewhat related to our finding of lower enforcement in committee chairs' home states, Agarwal et al. (2014) show that state regulators are more lenient than federal ones, which these authors partly attribute to regulatory capture. Others focus on the revolving door channel in the banking sector. For example, Braun and Raddatz (2010) find that banks that are politically connected are larger and more profitable than other banks, despite being less leveraged and having less risk. They suggest that banks may hire former politicians, in order to ensure favourable treatment. Adams (2013) finds that officers of larger banks are more likely to obtain Federal Deserve directorship; adding value in terms of positive average market reaction. Moreover, her findings reveal some evidence that banks with Federal Reserve directorships have lower probability to cease their operations in comparison to other banks within the industry. Using cross-state enforcement actions of regulators, Lucca et al. (2014) find that gross worker inflows into regulation from the private sector, and gross outflows from regulation are both higher during periods of intense enforcement; albeit the outflows are significantly smaller in magnitude. They conclude that their results are in favour of the regulatory schooling hypothesis rather than the quid-pro-quo explanation of the revolving door <sup>2</sup>. Shive and Forster (2016) reach a similar conclusion. They find that financial firms hire ex-employees of their regulators with an aim to reduce risk, and there is limited direct evidence of quid pro quo behavior in regulatory event frequency and fines.

The remainder of this paper is structured as follows. Section 2 provides an overview of the institutional background of U.S. Congressional Committees, U.S. Banking supervision and hypotheses development. Section 3 describes the sample data and variables. Section 4 provides an overview of the methodology employed and contemplates the results. Section 5 presents the findings of our further analysis and finally, Section 6 offers concluding remarks.

## 2 Institutional background and hypothesis development

# 2.1 U.S. Senate, House of Representatives and Congressional Committees

The U.S. Congress is the legislative branch of the U.S. Federal government and is composed of two chambers; the Senate (upper chamber) and House of Representatives (lower chamber). Senators and Congressmen (members of House of Representatives) are primarily chosen through direct election <sup>3</sup>.

The U.S. Senate is made up of 100 Senators (two for each state), who serve a six year

 $<sup>^{2}</sup>$ However, the example discussed by Fan et al. (2007) shows that the regulatory experience of former government bureaucrats does not necessarily make them more effective CEOs

 $<sup>^{-3}</sup>$  In some cases Senators may be chosen through gubernatorial appointment.

term. The Senate's powers and functions, as outlined by the Article One of the United States Constitution<sup>4</sup>, vary from "advice and consent" and approval of legislation suggested by the House of Representatives, to electing one of the top two recipients for Vice President's office in the case where no candidate receives a majority of electors. The House of Representatives consists of 435 voting members, each representing a congressional district and 6 non-voting members. Each member serves a two year term and there is no limit as to the number of terms an individual can serve. Each state is allocated a minimum of one member. The primary function of the House of Representatives is to initiate bills, which are then sent to Senate.

The U.S. Congress deals with a range of policy issues. The Committees of the U.S. Congress are legislative sub-organizations that are in charge of dealing with specific issues. This enables the efficient handling of such policy issues and it allows members to develop specialized knowledge on specific policy matters within their area of jurisdiction. A Committee's main tasks lie among monitoring government operations, identifying issues suitable for legislation, examining information on specialized issues and suggesting lines of action in their respective controlling body. The House has 23 Committees, whereas the Senate has 20. In regard to the Committee assignments (including Chair position), members are appointed by majority vote. For the chair position, seniority had traditionally been an important factor considered during the appointment procedure. Although all members exert influence on policy issues related to the committee's area of jurisdiction, the chair has considerable power to influence the policy-making procedure <sup>5</sup>. The powers of the Chairman lie among setting the legislative agenda, organizing meetings with the members of the committee, shaping debates related to particular issues, as well influencing the legislation brought to a vote.

<sup>&</sup>lt;sup>4</sup>For further information visit https://constitutioncenter.org/interactive-constitution/articles/article-i.

<sup>&</sup>lt;sup>5</sup>There is a growing body of literature that examines the role of Congressional Committees in policy making. For example, Berry and Fowler (2017) find that the majority of a committee's power is concentrated among chairs. Gropper et al. (2013) find that banks head-quartered in states where an elected official serves as a chair in their respective committee, tend to outperform banks in other states. Cohen et al. (2011) reveal that there is increased federal spending in the states where an elected official is a chair of an important committee.

As discussed earlier, our aim is to examine the level of influence of elected officials (member of House or Senate) in shaping bank regulatory decisions. There are two Committees closely related to the financial services industry which are taken into consideration in this study, namely the Committee of Banking, Housing and Urban Affairs (Senate's Committee) and the Financial Services Committee (House of Representatives' Committee)<sup>6</sup>. For example, the Banking, Housing, and Urban Affairs Committee engages in writing and passing legislation related, among other things, to banking, insurance, financial markets, securities and housing. The Committee also oversees the work of the Federal Reserve System, U.S. Department of Treasury, U.S. Securities and Exchange Commission and other financial regulators. Although all members of the Committee can exert, up to some extent, political power and influence; the Chair has considerable power to influence and shape regulatory issues related to the Committee's area of jurisdiction. Since Chairmen are also elected officials, one could expect the existence of a channel of influence flowing from the political environment within a state to the regulatory treatment of banks.

## 2.2 Enforcement Actions & U.S. Banking supervision

The US Banking sector operates under a dual banking system. This relates to the coexistence of federal and state banking systems. Three main regulatory agencies are responsible for the regulation and supervision US banks: (i) the Office of the Comptroller of the Currency (OCC) which is in charge of federally chartered banks (national banks), (ii) Federal Reserve Bank (FRB), which supervises state-chartered institutions that are members of the Federal Reserve System and (iii) the Federal Deposit Insurance Corporation (FDIC), which is responsible for federally insured depository institutions, which includes FED chartered banks that are not members of the Federal Reserve System, as well as state-chartered thrift institutions. For banks that are chartered on both a federal and state level, supervisory responsibilities are allocated amid the three regulatory agencies. According to the Dodd-Frank

<sup>&</sup>lt;sup>6</sup>The names of the House of Representatives Committee have changed, however the "Financial Services Committee" is the most up to date name.

Wall Street Reform and Consumer Protection Act; safety and soundness, deposit insurance, capital adequacy and systemic risk are considered to be the traditional components of US banking regulation. Regulatory agencies, therefore, monitor the safety and soundness conditions of the banks they supervise by conducting examinations on an on-site and off-site basis.

"Full-scope" on-site examinations are carried out by the regulatory agency in charge at least once a year <sup>7</sup>. Depending on the information collected through the above process, the agency assigns a rating, which reveals the financial condition and performance of the bank. The aforementioned procedure concerns the evaluation of the components of the Uniform Financial Institutions Rating Systems (UFIRS). This system is commonly referred to as the CAMELS rating system, which includes: Capital Adequacy (C), Asset quality (A), Management (M), Earnings (E), Liquidity (L) and Sensitivity to market risk (S)<sup>8</sup>. After the assessment procedure, a composite rating is assigned on each regulated institution and ranges among 1 to 5. Institutions assigned a rating of 1 or 2, appear to have a small weakness that may be controlled by its management or Board of Directors. A rating 3 or 4 suggests that there may be moderate to severe weaknesses encountered by the examined institution. In such cases, the managerial department may not be able or willing to promptly address the deficiencies that led to the weaknesses observed during the examination process. Finally, a rating of 5 reflects risky practices adopted by the institution and may lead to poor performance, as well as other severe problems. CAMELS ratings are strictly confidential, are not revealed to the public and are only available to each institution's senior management.

Off-site audits complement the CAMELS ratings from the on-site examinations and are

<sup>&</sup>lt;sup>7</sup>Federal Deposit Insurance Corporation Improvement Act, 1991.

<sup>&</sup>lt;sup>8</sup>Capital adequacy refers to the amount and quality of the institution's capital. Asset quality assesses the levels of existing and potential credit risk related to the institution's loan and investment portfolio. In addition, the management component, which is primarily a qualitative measure, refers to the effectiveness of internal control and audit systems; as well as the overall capability of board of directors and management to meet their roles. The earnings component rates the bank's earnings (current and expected). Liquidity refers to the assessment of the bank's ability to honour its cash payments as they fall due. Finally, sensitivity to market risk considers the management's capability to determine and control risks derived from the institution's exposure to its trade operations in the financial markets, as well as interest rate risk from non-trading positions.

implemented in order to monitor supervised institutions. In this case, information obtained by prior examinations and financial information provided in the Quarterly Reports of Condition and Income (Call Reports) is assessed. The above mechanisms are helpful in developing "early warning" models, which aid to identify institutions with riskier behaviour and thus, require supervisory attention. An important element to be highlighted at this point is that information derived by on-site and off-site programs, in combination with the CAMELS composite ratings assigned to each depository institution, play a key role in determining the type of enforcement action imposed on a bank.

There are two types of regulatory enforcement actions. First, informal enforcement actions are mainly voluntary commitments formed by the bank's management, board of directors or trustees when the level of misconduct is less severe. These actions are not publicly disclosed or legally enforceable and their main objective is to correct weaknesses identified during the supervisory process and reassure conformity with laws and regulations. Informal actions are imposed in cases where CAMELS ratings are of 3 or above. Common type of informal enforcement actions are commitments, Board resolutions, Memorandum of Understanding <sup>9</sup>. Second, formal enforcement actions are imposed in the case where misconduct is severe or in the case where informal actions previously imposed have not been adequate or effective. These types of actions are publicly disclosed and legally enforceable and are issued against institutions with CAMELS composite ratings of 4 or 5 or in some cases 3. The most common types of formal actions are termination of insurance, cease and desist orders, removal and prohibition orders, suspension orders and civil money penalties <sup>10</sup>.

One particular aspect in the context of our work is that enforcement actions must be "tailored" around an institution's individual characteristics and requirements. In general, enforcement actions target at correcting imperfections identified during the examination procedure and restoring an institutions financial health. The level of severity of an action depends on a variety of factors. For instance, overall financial performance, weaknesses,

<sup>&</sup>lt;sup>9</sup>OCC PPM, 5310,-3 at 8-9, 2011; FDIC, 2014, II - 8.1.

<sup>&</sup>lt;sup>10</sup>FDIC, 2014, II - 8.1; OCC PPM, 5310,-3 at 8-9, 2011.

attitude of an institution's bank management in committing play an important role on the decision of the type of formal enforcement action to be imposed. Moreover, composite CAMELS ratings play a crucial role the this decision. Finally, prior non-addressed problems or shortcomings can lead to the implementation of formal enforcement actions , or even in some cases additional formal actions.

## 2.3 Hypothesis development

The research objective of the present study is to examine whether banks headquartered in a "politically important state" are more likely to receive more favourable regulatory treatment. Within this context, a "politically important state" refers to any state where an elected official (Senator or Congressman) holds a chair position within a congressional committee related to the banking and financial services industry. It is important to note that Senators and Congressmen very often visit their home states in order to meet with citizens and/or other systematically important parties, like banking institutions. During such interactions, banks can collectively or individually exert influence towards congressional officials and can gauge their attention more conveniently to regulatory and supervisory matters. In the meantime, banks adopt various corporate political strategies, such as spending money on campaign contributions, in order to further influence and support the congress and bureaucrats. In turn, members of Congress listen to the issues raised during these meetings and have the power to raise these on a higher level. This is even more pronounced if the elected official holds a chair position in a congressional committee, which is in fact one of the principal reasons for which we focus on the political influence of the chair.

Reasonably, the above can be beneficial for a bank, depending on the intentions of the elected official. For instance, the politician may be a "social planner", thus, aiming to maximise social welfare. Under this view, the politician will act in such ways as to promote the proper implementation of regulations, reforms and enforcement mechanisms. Alternatively, the politician may be "selfish", in terms of catering for his/her own interests, aiming to maximise his/her own personal agenda, (i.e. the possibility of re-election). Under this view, the politician will provide accommodating treatment for the banks in his/ her home state. This accommodating treatment could include delaying or avoiding the closure or punishment of financially troubled institutions, or the implementation of less stringent regulation for a particular group of banks. In exchange, banks provide their support through campaign contributions, lobbying or even future employment.

Moreover, a prior study by Correia (2014) attributes the political control of regulatory agencies to two distinct theories: The Iron Triangles and the Congressional Dominance theory. The former states that a mutually beneficial, three-way relationship exists between agencies (such as regulators), special interest groups and congressional committees. Under this view, congressional committees offer political support and funding to agencies, which provide a lax regulatory environment or specific favours to special interest groups. Special interest groups support (e.g. through votes, monetary contributions) congressional committees, who in turn have the power to pressurize agencies to implement favourable treatment or policies towards the special interest groups. The latter theory suggests that a principalagent problem arises between congress and agencies. In particular, it suggests that Congress assigns various tasks and formulates alternative monitoring systems in order to motivate agencies to act on their behalf, i.e. to maximize political support.

In addition, there is some evidence of a geographic component of firms' political influence and, more specifically, that firms located in a politician's home state enjoy preferential treatment. Faccio and Parsley (2009) measure the value of this political influence to companies by showing that the unexpected termination of this influence (e.g. because the politician resigns) is associated with a 1.7% reduction in the company's market value. Preferential treatment may be manifest in the form of higher federal investment in that state generally (Cohen et al., 2011), or banks from this state being less likely to fail despite higher leverage (Kostovetsky, 2015) or more likely to receive government funding (Duchin and Sosyura, 2012). In that last context, Chavaz and Rose (2016) suggest that the preferential treatment might have been motivated by the expectation that banks which benefit from such treatment will adopt investment policies that are more in line with the politician's preferences, but other motives such as the expectation that this behaviour be rewarded by voters, or politicians being more likely to have personal or financial ties with local banks also seem plausible.

Given the above discussion, the main hypothesis to be tested is the following:

H1. Banks that are headquartered in a politically important state are more likely to receive favourable treatment by their supervisors, in the form of lower probability of receiving an enforcement action.

## 3 Sample, data and Variable selection

The working sample of the present analysis consists of a fully hand-collected dataset including information for U.S. Commercial banks for the years 2000q1 - 2015q4. Financial characteristics of banks are obtained by the 031/341 Call-report files available from the Federal Financial Institutions Council (FFIEC) and Chicago Fed website. The sections below provide an overview of the construction procedure of the final working sample, the variables included in the analysis, and their descriptive statistics. Table 1 provides a list of the variables.

#### [Insert Table 1 Around Here]

### 3.1 Enforcement action variables

Data on enforcement actions are retrieved from the three federal regulatory agencies' websites; namely Federal Reserve Bank (FRB), Federal Deposit Insurance Corporation (FDIC) and the Office of Comptroller of the Currency (OCC).

The types of actions taken (in order of severity) are: Deposit Insurance Termination,

Cease and Desist orders, Formal Written Agreement/ Supervisory Agreement, Prompt Corrective Action/ Capital Directive, Civil Money Penalty, Call Report Penalty, Penalties on violations of specific laws. A detailed description of each type along with their classification, can be found in Appendix I. Enforcement actions are split into two categories: severe and less severe. While we initially consider all types of actions, we subsequently focus on the severe ones, i.e. Deposit insurance termination, Cease and Desist orders, Formal Written Agreements and Prompt Corrective Actions. There are various reasons for this. First, severe actions are more closely related to safety and soundness issues of banks. In contrast, less severe actions are usually issued against institutional affiliated parties and are therefore, not related to deficiencies observed over the financial condition of an institution. Moreover, Delis et al. (2016) show that such actions do not have an impact on a bank's financial condition. Finally, our preliminary results show that these types of actions are not associated with political influence.

In order to match the enforcement action data to the Call Report Data, we follow two steps. First, in cases where the unique identifier number of the punished institution (cert or rssd9001) is available, this is directly matched with the one provided by the Call Report data. Second, if this information is not provided, we match the name, city and state of each institution. Since Call Report data are provided on a quarterly basis, enforcement actions are also matched quarterly. For example, if Bank XX received an enforcement action during January 1st- March 31st, then this information is assigned to the 1st quarter. In the case where a bank received multiple types of enforcement actions during a given period; the most severe type is retained.

A dummy variable ("*Action all types*") is constructed, which takes the value of 1 if a bank received a formal enforcement action of any type (severe or less severe) in a particular quarter and the value of 0 otherwise. We construct two additional variations of this dummy variable. The first ("*Severe action*") takes the value of 1 in the case of severe actions, and the value of 0 otherwise. Similarly, the second ("*Less severe actions*") takes the value of 1

in the case of less severe actions, and the value of 0 otherwise. The bank-quarter number of all types of enforcement actions in the sample is 3,925. Table 2 provides the descriptive statistics of the enforcement actions of the sample. Overall, there is a general increase in the total number of enforcement actions issued around the crisis years. As it concerns the severe type of enforcement actions, which are the core focus of the present study, it appears that Formal Written agreements and Cease and Desist orders are the most common types of severe actions imposed; whereas the least frequent is Deposit Insurance termination. In regard to the actions issued per agency, FDIC appears to have issued the most severe actions in comparison to the actions issued by the FED and the OCC.

[Insert Table 2 Around Here]

## 3.2 Political influence variable

Information related to Committee Chairmanships is obtained by the relevant committee's website: U.S. Senate Committee on Banking, Housing and Urban Affairs <sup>11</sup> and U.S. House of Representatives Financial Services Committee <sup>12</sup>. Biographical information of Committee Chairmen is obtained by the Biographical Directory of the United States Congress <sup>13</sup>. The timeframe of the analysis addresses the period between 2000q1 - 2015q4; thus, covering 9 Congresses (106th -114th). Table 3 provides a summary of the available information of banking committee chairs for the sample.

### [Insert Table 3 Around Here]

As mentioned previously, the present study focuses on two particular Committees that specialize on policy issues related to banking and financial services. These are the Committee of Banking, Housing and Urban Affairs (Senate) and the Financial Services Committee (House of Representatives). We do so because their area of jurisdiction is closely related to

<sup>&</sup>lt;sup>11</sup>https://www.banking.senate.gov/public/index.cfm/home.

<sup>&</sup>lt;sup>12</sup>https://financialservices.house.gov.

<sup>&</sup>lt;sup>13</sup>http://bioguide.congress.gov/biosearch/biosearch.asp.

the banking sector and thus, we expect that it is more likely that they would be approached by banks in order to discuss or settle any issues. In the core analysis of our study we focus on elected officials that hold a chair position on these Committees. The rationale of our approach is that the chair of a committee exerts superior influence than that of any other member. However, in a robustness check we will extend the scope of the political variable by including all members of both committees.

In order to capture political connections, we employ a binary variable ("*Committee chair*"), which takes 1 if a bank's parent company operates in a state where an elected official (Senator or Congressman) holds a chair position in their respective committee in a given year, and 0 otherwise. In the working sample, the number of connections identified is 18,603 (4.94%) bank-quarter observations over the total 376,358 bank-quarters for the period 2000-2015. From the summary information reported in Table 3, it appears that for the Senate committee, there are five different chairmen present in the sample, representing Texas, Maryland, Alabama, Connecticut and South Dakota. As for the House Committee, there are four different chairmen, who represent the states of Ohio, Massachusetts, Alabama and Texas. In terms of party affiliation, there are six Chairmen representing the Republican party and four representing the Democratic party. The Chairmen's experience while in service ranges from 11 to 33.5 years (on average), whereas their age ranges from 57 to 82 years (on average).

## 3.3 Financial variables

As outlined in Section 2.2., the CAMELS ratings play a crucial role, for regulatory purposes, in assessing the financial condition and performance of banking institutions. However, these ratings are confidential and we therefore have to use individual financial ratios proposed in the literature. Cole and White (2012) show that such traditional proxies for CAMELS components do an excellent job in explaining the bank failures during 2009 as they did in the 1985-1992 crisis. Our approach is consistent with the one followed in past studies (Lambert, 2019; Delis et al., 2016). Moreover, we control for bank size to account for differences in terms of organizational complexity, too-big-to fail issues, and that large firms may have more political influence (Schiffer and Weder, 2001).

In particular, we capture capital adequacy by the risk-based capital ratio. To control for asset quality, and in particular credit risk, we use the risk-weighted assets ratio. We also use the return on assets ratio as a measure of earnings and managerial ability. To proxy for the liquidity component we use the ratio of liquid assets to total assets. Finally, we include the natural logarithm of total assets to account for bank size. A detailed description of these variables is available in Table 1. Financial variables have been trimmed in the (-3,+3) standard deviations window around their mean.

Table 4 provides the descriptive statistics of the above variables. Panel A includes information regarding financial and political influence variables for the full sample (part i), political connected banks only (part ii), punished banks (part iii) and punished and connected banks sample (part iv). Panel B presents the descriptive statistics for macroeconomic and other state-level variables. Table 5 reports the mean difference of all variables for two groups: non-punished banks and those punished with a severe enforcement action. There is divergence in the mean values of the variables across samples in Table 5. According to the figures reported, it is apparent that punished banks are of lower quality in terms of capitalization, asset quality and earnings. However, it appears that punished banks are more liquid than non-punished banks. Moreover, punished banks appear to be larger. Focusing on the political influence variable, it appears to be higher for non-punished banks and this difference is statistically significant at the 1% level. At this point there is preliminary evidence of the fact that politically connected banks are less often enforcement targets. The following section provides a multivariate analysis in order to examine the relationship more thoroughly.

[Insert Tables 4, 5 Around Here]

## 4 Analysis and results

### 4.1 Baseline results

To examine the relationship between political influence and the likelihood of an enforcement action, we follow prior literature and use a non-linear probability model. Therefore, as in Lambert (2019), we estimate our baseline equation with a Probit model of the following form:

$$Action_{i,k,t} = \alpha + \beta_1 Pol.Influence_{i,k,t} + \beta_2 Controls_{i,k,t-1} + \gamma state_k + \delta quarter_t + \varepsilon_{i,k,t}$$
(1)

where *i* refers to bank *i*, *k* refers to state *k* and t refers to quarter *t*. Action<sub>i,k,t</sub> is a dummy variable that takes 1 if a bank *i* received an enforcement action in quarter *t* and 0 otherwise. Political Influence<sub>i,k,t</sub> is a dummy variable that takes 1 if bank *i*'s parent company operates in a state where an elected official holds a chair position in a Congressional Committee related to banking and financial services in quarter *t*, and 0 otherwise. Therefore,  $\beta_1$  is the coefficient of the variable of interest and reveals the effect that politically important states have on the probability of receiving a severe type of enforcement action. Controls<sub>i,k,t-1</sub> is a vector of covariates accounting for bank-level characteristics for bank *i* at quarter *t*-1<sup>14</sup>. Given the nature of our setting it is important to account for the observable and unobservable characteristics across states. We therefore include a full set of state dummies. In order to account for variation across time, we additionally include a set of quarter dummies. Equation (1) is estimated using robust standard errors clustered by bank.

Table 6 provides the estimation results of our baseline model. Column 1 reports the results for all types of actions, Column 2 reports the results when severe types of actions

<sup>&</sup>lt;sup>14</sup>We employ financial variables lagged by one quarter to account for the fact that financial statements are published with a lag.

taken into account, and Column 3 reports the results when less severe types of actions are considered. The results in this table are the output of Equation (1) being estimated using a Probit model specification. We report both the coefficient estimates as well as marginal effects at means in order to facilitate interpretation.

The coefficient of the core variable of interest,  $PoliticalInfluence_{i,k,t}$  is negative and significant at the 1% level for severe types of actions, confirming our earlier observations from Table 5. This suggests that banks whose parent company operates in a politically important state are less likely to be subject to severe types of enforcement actions.  $PoliticalInfluence_{i,k,t}$ has an insignificant coefficient when less severe type of actions are included. Thus, the effectiveness of our political influence measure appears to be important for cases where banks are subject to enforcement actions that are more closely related to more severe types of misconduct. Our results also convey economic significance. Focusing on the severe type of actions, political influence reduces by 0.1 percentage points the probability of a bank being punished with a severe type of action. Moreover, given that the unconditional probability of a bank receiving a severe type of action being equal to 0.5%; we can infer that political influence reduces by almost 20% the probability of a bank receiving a severe type of action. Although this figure is somewhat lower than the one reported by Lambert (2019) (52.1%), it does still indicate a pronounced impact of political influence on regulatory enforcement likelihood. A potential reasoning behind this difference could be that influence from lobbying involves a bank itself exerting direct pressure to the regulator or politician; whereas this directness is absent when examining political influence stemming from the politician to the banking industry.

We further estimate our baseline model by making use of a Linear Probability Model using fixed effects. We present the results in Column 1 (all sections), Column 2 (severe actions only), and Column 3 (less severe actions only) of Table 7. The findings suggest that when controlling for fixed effects at the bank level, the coefficient of the core variable of interest, remains negative and statistically significant. In addition to the Linear Probability Model, we also estimate the baseline model implementing a suitably modified logit model (Rare events logit model) that accounts for the fact that enforcement actions can be seen as "rare events". King and Zeng (1999) highlight that models with binary dependent variables, like logit and probit can underestimate the probability of rare events. Descriptive statistics in Tables 2 and 4 reveal that enforcement actions can be indeed seen as rare events. Taking together both types of enforcement actions, such actions represent about 1% of the total of bank observations, whereas when we focus on severe actions only, this percentage drops to 0.50%. Therefore, as in Lambert (2019), we perform a robustness test using a rare events logistic regression approach(King and Zeng, 1999, 2001; Tomz and Zeng, 1999). The output of this exercise provides similar results, which are reported in Table 8. In addition to the above two models, we also estimate the baseline model using Firth's Penalised Maximum Likelihood Estimation (PMLE) method, which is useful in reducing prediction bias <sup>15</sup>. As evident from table 9 the coefficient of the core variable of interest remains intact.

## [Insert Table 6 Around Here]

To examine further the sensitivity of our results to the use of additional models, we reestimate Equation (1) under four assumptions <sup>16</sup>. More specifically, we: (i) use a Binary Logit model, (ii) use a multinomial logit model that accounts simultaneously for all three possible outcomes (i.e. severe, less severe, and no enforcement action), (iii) use ordered logit and probit models that take into account the ordering of the severity of the action, and (iv) estimate a binary model where the dependent variable takes the value of "1" only if the bank received a "severe" enforcement action in that period. As discussed earlier, the dependent variable takes the value of 1 in the case of severe actions and 0 otherwise. Thus, the value of zero has been assigned in cases of both less severe actions and no enforcement actions of any type. At this stage, we drop the cases of less severe actions from the analysis and essentially re-estimate the model with the dependent variable taking the value of 1 for severe actions

<sup>&</sup>lt;sup>15</sup>For further information please see Firth (1993).

<sup>&</sup>lt;sup>16</sup>To conserve space we do not present these estimations. They are available from the authors upon request.

and the value of 0 for no enforcement action of any type.

In all the four cases discussed above, our findings remain the same and consistent with the ones presented in Table 6. Therefore, in the analysis that follows we concentrate on severe-type actions only, an approach that is consistent with Lambert (2019). Lambert (2019) provides two further justifications that could possibly also explain why less severe actions appear to be insignificant. First, less severe actions are usually issued against individuals affiliated with an institution and not because of the financial condition of the institution. Second, less severe actions do not have a direct impact on bank activities.

As a further test, we use alternative variations of the control variables, by either using contemporaneous versions or their 1st differences. Finally, Equation (1) is re-estimated with the use of alternative bank-specific control variables. For example, instead of the risk-based capital ratio, the base model is estimated using the equity to assets ratio as a measure of capitalization. Another example is the use of the natural logarithm of deposits, which indicates the size of the bank's sources of stable funding. These controls are included one by one as well as jointly in these set of regressions. We also estimate the baseline model using alternative measures of credit risk, such as non-performing loans or loan loss provisions ratio. Our findings are robust to these variations. In all the cases, the coefficient of our political influence variable remains negative and statistically significant. We do not report these results, however they are available upon request.

In terms of the remaining control variables, the findings appear to be in line with the ones of prior studies. Briefly, it appears that on average, punished banks are larger, less capitalized, have lower risk weighted assets, return on assets, and higher liquidity  $^{17}$ .

In the following section (Section 4.2) we examine the sensitivity of our results while (i) restricting the sample, (ii) using additional control variables, and (iii) addressing endogeneity concerns. We discuss the results in the sub-sections that follow.

<sup>&</sup>lt;sup>17</sup>The findings about size and liquidity may appear to be counterintuitive at first. However, they are consistent with the findings of earlier studies. For example, Lambert (2019) also finds that liquidity and bank's size enhance the likelihood to receive an enforcement action. Delis et al. (2016) also report that punished banks have higher liquidity ratios relative to their sample average

[Insert Tables 7, 8, 9 Around Here]

## 4.2 Robustness analysis

#### 4.2.1 The impact of the Financial Crisis

To begin with, we explore the potential influence of the financial crisis on the findings of the present study. As depicted in the descriptive statistics (Table 2), the number of formal enforcement actions imposed during the crisis years reached record high. The question that arises is whether political influence during the years of the crisis remains the same. To explore this, we follow empirical precedent (Lambert, 2019) and re-estimate the base model while restricting the analysis to the years of the crisis, i.e. 2008-2009. We also explore the impact of political influence during non-crisis periods, and thus, re-estimate the base model while dropping observations from years 2008 and 2009. Columns 1 and 2 in Table 10 report the results obtained. The overall findings suggest that for both crisis and non-crisis years, the coefficient of the political influence measure remains negative and statistically significant. However, it is worth mentioning that the level of statistical significance deteriorates at the 5%, with the coefficient of the political influence variable for the crisis only sample being marginally towards the upper limit.

### 4.2.2 Controlling for systematically important and Highly capitalised banks

A second set of tests is carried out examining certain sub-samples of the core working sample. In particular, findings so far could be driven by a specific group of banks due to their financial characteristics. The sensitivity tests implemented in this section aim to address two main concerns. The first relates to the fact that top performing, as well as poor performing banks could potentially be influencing the estimation results. Thus, we first estimate Equation (1) by excluding the strongly capitalised banks of the sample (top 1%) and then by excluding the poorly capitalised banks of the sample (bottom 1%). The second concern relates to the fact that some banks may be receiving preferential treatment because they are considered systematically important. In order to control for this, we exclude large banks from our sample -in particular the top 1%- and re-estimate our base model. In particular, we estimate the baseline model by (i) excluding largest 1% banks per state and (ii) largest 1% banks per year. As reported in Table 10 the core variable of interest retains its sign and significance <sup>18</sup>.

[Insert Table 10 Around Here]

### 4.2.3 Alternative channels of influence

In this section we aim to control for alternative channels of influence. In particular, given that prior literature (Lambert, 2019) has shown a significant relationship between lobbying and probability of enforcement, we focus on the lobbying activity of banks. To do this we hand collect information regarding bank lobbying activity provided by the Center of Responsive Politics (CRP). We use the information obtained to create a dummy variable that takes '1' if a bank is engaged in lobbying activities in a particular year, and '0' otherwise. It is important to note at this point that the majority of lobbying reports are filed by parent companies. We therefore, follow prior literature and assign any parent lobbying activity to each subsidiary. Column 1 in Table 11 reports the results obtained when re-estimating our baseline model by additionally controlling for bank lobbying, in order to account for alternative pressures of influence. The main findings suggest that even when controlling for lobbying, our political influence variable remains statistically significant at the 5% level.

<sup>&</sup>lt;sup>18</sup>It is worth noting that when excluding high capitalised banks from the sample, the capital ratio appears to be more negatively associated with enforcement likelihood, whilst the opposite holds when excluding low capitalised banks. A potential explanation for this could be that when high capitalized banks are excluded from the sample, the remaining banks are those with low and medium capitalisation. Thus, for this particular group of banks the availability of capital forms an important factor that decreases enforcement likelihood. The opposite holds when excluding low capitalised banks. That is, the remaining sample of banks includes medium to high capitalised banks. In this case, the availability of capital is less important in regard to the likelihood to be assigned an enforcement action

#### 4.2.4 Differences among regulatory agencies and distance to regulator

First, we begin by including two dummy variables to account for potential differences between the three regulatory agencies <sup>19</sup>. The first dummy takes the value of '1' in the case of banks regulated by the FDIC and the value of 0 otherwise; whereas the second takes the value of '1' in the case of banks regulated by the OCC and the value of 0 otherwise. Banks regulated by the FED form the omitted category. The results in Table 11, Column 2 reveal that both variables enter the regressions with a positive and statistically significant coefficient. Thus, banks regulated by the FDIC or the OCC are more likely to be subject to an enforcement action than banks regulated by the FED. The inclusion of these variables in the analysis does not alter our key finding. It is worth noting that, as shown in Table 2, the number of severe actions varies among regulators. There are various potential explanations for this. First, the number of banks under the supervision of each regulator also varies. For instance, according to a recent report by Deloitte (2015) the FDIC had jurisdiction over 4,138 commercial and savings institutions as of December 31, 2014, compared to 1,513 and 858 institutions under the purview of the OCC and the Federal Reserve System, respectively. Second, it could reflect differences in the regulatory approaches, the supervisory mandates, and the type of banks under supervision. For example, the same report by Deloitte (2015) points out that the composition of enforcement actions issued by each regulator (e.g. formal agreements vs cease desist orders) between 2000 and 2014 suggests that the FDIC takes a more direct approach against institutions and the Federal Reserve System takes a less direct approach, with the OCC having a more balanced mix of severe actions. Third, it is possible that these differences are related to different political connections. Therefore, in unreported regressions

<sup>&</sup>lt;sup>19</sup> Ioannidou (2005) finds that the Fed's monetary policy responsibilities alter its bank supervisory behaviour, as captured by formal actions. More detailed, she finds that a tightening of the Fed's monetary policy is associated with a lower probability of intervention. However, the monetary policy actions do not affect the behaviour of the other two agencies (i.e. FDIC, OCC). Agarwal et al. (2014) also conclude that regulators can implement identical rules inconsistency due to their institutional design and incentives. Their comparison of federal and state supervisory ratings within the same bank reveals that federal regulators are systematically tougher; however, they do not find evidence that this is due to regulator self-interest and revolving doors.

we estimate separate regressions for the two types of regulators for which we have adequate data (i.e. OCC and FDIC). In both cases the results hold <sup>20</sup>.

In addition to the above, we also opt to control for the distance between a bank's headquarters from its regulator. In particular, we include the variable 'Distance to regulator', which includes the distance in km between a bank's headquarters and nearest Federal Reserve Bank (Main office or Branch). Prior research (e.g. Lim et al., 2016) provides evidence suggesting that indeed a bank's distance from a regulator's field office has an impact on regulatory monitoring; since distance can reflect the level of information asymmetry between the regulator and the bank. Taking that into consideration, we re-estimate our baseline model by controlling for this type of information asymmetry and find that our key variable of interest remains intact.

#### 4.2.5 State-related analysis

In this section we aim to control for state-related characteristics that could be driving our results. First, we estimate our baseline model by clustering standards errors at the state level, instead of the bank level and this does not alter the results obtained thus far.Our second exercise is related to the concern that banks headquartered in states, which are considered to be "important banking states" may be driving the results of our study. We thus, perform an analysis where we exclude banks headquartered in New York, California and North Carolina. The results reported in Column 4 ,Table 11 suggest that even when controlling for the "important banking states<sup>21</sup>" our core variable of interest is negative and statistically significant.

### [Insert Table 11 Around Here]

 $<sup>^{20}</sup>$ In the case of the Federal Reserve System, there are not enough data for a meaningful analysis. For example, there are only 56 cases of severe actions over the entire period of our analysis. At the same time, among these 56 cases, there are only 2 cases where the variable of interest (i.e. Committee Chair) takes the value of 1.

<sup>&</sup>lt;sup>21</sup>For instance, according to the Global Financial Centers Index Report (2018), which examines statistics on the leading financial centers on a global scale, New York (NY) and San Francisco (CA) are ranked amongst the top 25 cities. Full report can be found here: https://www.longfinance.net/programmes/financial-centre-futures/global-financial-centres-index/.

## 4.3 Addressing selection bias

In this section we perform further analysis to address potential issues related to selection bias and to confirm the draw of causal effects in the results our study. In particular, we employ this approach as we aim to rule out the potential that the observed inverse relationship between political influence and enforcement likelihood is driven by specific bank characteristics, such as size, profitability, etc. We address this issue by implementing a Propensity Score Matching approach, which is commonly applied in relevant literature (List et al., 2003; Borghesi and Chang, 2015; Lambert, 2019; Papadimitri et al., 2019a). This approach is suitable in addressing causal treatment effects and its applicability is of use when there is a relatively small amount of the treated group (i.e. politically connected banks), as opposed to the non-treated group (i.e. non-connected banks). In addition, it facilitates to ensure that any selection bias is not responsible for the observed relationship of interest.

The core element of this approach is the calculation of a propensity score or else, the predicted probability, which is defined as "the probability of treatment assignment conditional on observed baseline characteristics" (Austin, 2011). The core focus of this approach is to obtain a probability for which a particular group within the sample will be treated or in other words, participate in an event, given a set of common characteristics for that group. Therefore, the analysis in practical terms involves the categotization of individuals of the sample's population in accordance with certain characteristics which are observable. Then based on those characteristics, a probability is calculated in order to establish whether the pattern originally observed in the main model holds or not. In such context there are two main types of groups: (i) treated group - i.e., the group that participates in a particular event and (ii) non-treated group - i.e., the group that does not participate in a particular event. Once the categorisation takes place it is possible to calculate the difference in means among the treated group (Average Treatment Effect for the treated).

In the current setting of this paper, in order to carry out the analysis, a Probit model is

estimated where the dependent variable is the treated variable (i.e. the political connections variable) and the independent variables include a set of bank-level characteristics which determine the matching of the two groups. In particular, we consider the risk-based capital ratio, risk weighted assets, liquidity ratio, return on assets ratio and size. The propensity score is then calculated. Table 12 reports the results obtained. Column (1) reports the Average Treatment Effects for the Treated (ATT), whereas Column (2) reports he number of matches obtained. The matching technique implemented is 'Near neighbor' for n=1,10,50,100 and caliper value equal to 0.01. The overall results suggest that politically connected banks are less likely to be punished and most importantly, highlight that there is a significant difference among connected and non-connected banks.

[Insert Table 12 Around Here]

## 5 Further Analysis

Thus far, our findings suggest that elected officials holding a powerful position within a congressional committee that oversees the financial services industry, can influence the probability of a bank receiving a severe type enforcement action. In this section we complement our analysis by exploring alternative explanations and potential channels that could provide further insights.

## 5.1 Committee and politician related factors

#### 5.1.1 Politician's power and influence

We attempt to explore personal traits of main committee chairmen in order to assess whether individual characteristics play a role in our empirical investigation. Prior studies (e.g. Gropper et al., 2013, 2015; Vidal et al., 2012) have highlighted the importance of politician seniority. In fact, Gropper et al. (2013), find that a chairman's age have a significant effect on bank performance. We therefore estimate our baseline model by examining whether long/short seniority lead to higher/lower probability of enforcement action likelihood. We explore an elected official's level of seniority through the combined average age of two chairmen (House and Senate Committee) in a particular year. This measure captures additional dimensions of power and knowledge of the industry from prior employment. A person's age can also reveal and drive other aspects, such as their decision making, level of risk or even career concerns.

Following Gropper et al. (2013), we define quarters in which the average age of both committees' chairmen is above the overall average <sup>22</sup> in the sample as "high seniority" <sup>23</sup>. We then run our baseline regression separately for the subsamples of periods with high and low seniority <sup>24</sup>.

Table 13 reports the results. Our findings confirm our main results for years where committee chairs were more senior than the average, at the 1% level of statistical significance. For periods with junior committee chairs, the coefficients of our political influence variable remains negative, but it is now insignificant. This finding highlights the fact that seniority forms a key contributing factor when it comes to the extent of influence exerted by the politician in regard to enforcement decision making. Therefore, the power, knowledge and potentially the relationships acquired over the years of age and in service appear to enhance the influence of politicians <sup>25</sup>.

### [Insert Table 13 Around Here]

 $<sup>^{22}</sup>$ We also construct another version of this variable using the median as the cut-off point as an alternative robustness check. The main results hold. These estimations are available upon request.

<sup>&</sup>lt;sup>23</sup>To provide an illustrative example, consider the state of Florida, which during the years examined in our study has not had an elected official holding a chair position within a powerful committee overseeing the Financial Services industry. However, with the use of the independent sampling banks from Florida can be considered in our sample in terms of high/low seniority, depending on the bank's chair age from another state, for different time periods. Thus, in this case, there is no variation across states during a particular point in time; however there is variation across time and therefore this approach provides a ground to test the level of impact of individual characteristics.

 $<sup>^{24}</sup>$ Due to the complications when using interactions in nonlinear models pointed out by Ai and Norton (2003), we resort to examining the impact of the individual characteristics on our main effect by splitting the sample according to these characteristics. Estimating the interactions in a linear probability model yields qualitatively similar results.

<sup>&</sup>lt;sup>25</sup>We also examined the impact of experience specifically in the role of chairmen but did not find a difference in the chair effect between high and low levels of experience.

In our attempt to further explore the potential motives of elected officials holding powerful positions in Congressional committees in promoting laxer treatment of banks headquartered in their home state; we examine two alternative paths. First, we consider the total number of votes received by a chairman as a portion of the total number of votes within the district or state (depending on chamber). We thus explore an elected official's "popularity" within their state or district, by considering the percentage of votes they receive for each election cycle. This could be a conditional factor for whether the politician was easily elected or whether the amount of votes received could potentially translate in favors owed and in more general terms, motivate elected officials. Prior studies have shown that there exists a connection between firms' political connections and job and plant creation rates, which is more pronounced in areas where elections are highly contested (Bertrand et al., 2018). We collect voting information from the Federal Elections Commission data on Elections and Voting <sup>26</sup> and manually match this information to each elected official of our sample.

Second, we collect information for the total dollar amount of contributions received by each elected official in our sample. Our decision to employ this measure is driven by the fact that campaign contributions are considered to be an indirect measure of a politician's power (Berry and Fowler, 2017). To tailor the investigation around our research question, we focus on contributions particularly from the Financial Services industry and use the ratio of the sum of contributions received from the financial sector as a fraction of the total contributions received by a candidate. We choose this measure to specifically observe the level of "aggressiveness" of financial institutions in approaching politicians<sup>27</sup>. We obtain this information from the Center of Responsive Politics (Open Secrets) website.

Once the above information is collected, we create two main measures: (i) %Votes and (ii) %ContributionsFIN. Following the same procedure described above, we define quarters in

 $<sup>^{26}</sup>$ Further information on the data can be found here: https://transition.fec.gov/pubrec/electionresults.shtml.

<sup>&</sup>lt;sup>27</sup>Figures reported by the "Donation Concentration Metric" study, which examines the concentration of donations during an elected official's career, show that for the banking industry, four (Spencer Bachus, Jeb Hanserling, Richard Shelby, and Tim Johnson) out of ten have been Chairmen of an important Committee in Finance and are also part of our sample in our study.

which the average ratio of contributions (votes) received by both committees' chairmen from the financial sector to total contributions is above the overall average as "high contribution" ("high votes", and quarters in which average contribution ratio (votes) of both committees' chairmen in a given year is below average as "low contribution" ("los votes"). The results of our baseline regressions for the subsamples of high versus low contributions and high versus low votes are reported in Table 13.

Our findings reveal that for the sub-samples where chairmen received higher votes or high contributions, the political influence variable is negative and statistically significant. In contrast, in the case of the "Low" sub-samples, the coefficient of the core variable of interest is insignificant. Therefore, from a general and aggregate scope, it appears that higher popularity (in terms of votes received), as well as higher amounts of money received by the financial sector, enhance politician's effectiveness and potentially motivation, on reducing the likelihood of a bank being punished <sup>28</sup>.

## 5.1.2 Political influence, committee membership and sub-committee chairmanship

The analysis presented so far has focused on the role of the Chairmen. In this sub-section we extend our analysis to examine the role of other members of the Committees. We obtain information on members of the Senate's Committee on Banking, Housing and Urban Affairs and the House of Representative's Financial Services Committee from 103rd to the 114th Congress from the Charles Stewart III and Jonathan Woon data set <sup>29</sup>. Due to the fact that it is very likely for more than one committee member to be elected from a particular state, we aggregate the information from the Charles Stewart III and Jonathan Woon data set and

<sup>&</sup>lt;sup>28</sup>We also explore the potential effect that political party affiliation has on enforcement action likelihood. We implement two approaches. First, we estimate the baseline model using a dummy variable representing the chairman's political party (i.e. democrat or republican). Second, we explore political party alignment among the chairmen and president. The findings show no significant relationship and thus are not reported for brevity. They are, however, available upon request.

<sup>&</sup>lt;sup>29</sup>Please find further information on the data and related studies at the Charles Stewart Congressional data page http://web.mit.edu/17.251/www/datapage.html.

construct a dummy variable (*Committee member*), which takes the value of 1 if a bank is headquartered in a state where there is at least one elected official that serves as member in one of the two Committees of interest, and the value of 0 otherwise.

Additionally, we explore the potential influence of officials serving as Chairs in subcommittees. As mentioned earlier, congressional committees are in charge of working on specific tasks and issues related to a particular area of jurisdiction. Thus, the main committees consist of a number of sub-committees, each of which is in charge of dealing with even more specific tasks within the jurisdiction of their "parent" committee. For example, the House "Financial Services" Committee currently consists of five sub-committees (Capital Markets, securities and Investment; Financial Institutions and Consumer credit; Housing and Insurance; Monetary Policy and Trade; Oversight and Investigation; Terrorism and Illicit Finance), which engage in and deal with a set of very specific issues. In terms of subcommittee organization and structure, similarly to the main committees, the subcommittees also are assigned a chairman and a group of members who are elected officials across different states <sup>30</sup>.

For the purposes of our study, we focus on the following two sub-committees, which are considered to be the more relevant to our analysis: (i) The House of Representative's subcommittee "Financial Institutions and Consumer Credit" and (ii) The Senate's subcommittee on "Financial Institutions and Consumer protection"<sup>31</sup>. We obtain historical information on sub-committee chairmanship from the Center of Responsive politics website from the 106th -114th Congress and construct a binary variable (*Subcommittee Chair*), which takes the value of 1 if a bank is headquartered in a state where an elected official holds a Chair position on one of the two aforementioned sub-committees and the value of 0 otherwise.

<sup>&</sup>lt;sup>30</sup>Further information on subcommittees could be found in the following: (i) Senate's Banking, House and Urban Affairs Committee can be found: https://www.banking.senate.gov/about/subcommittees ; (ii) House of Representative's Financial Services Committee can be found https://financialservices.house.gov/about/subcommittee-membership.htm.

<sup>&</sup>lt;sup>31</sup>For example, the jurisdiction of the "Financial Institutions and Consumer protection" subcommittee lies -among others- around banking institutions, deposit insurance, regulatory activities of the Fed, as well as those of the OCC and FDIC.

We re-estimate the baseline model with the inclusion of the "*Committee member*" and "*Subcommittee Chair*" variables. The results are reported in Table 14. Our findings reveal a statistically significant and inverse relationship between the probability of a bank receiving a regulatory enforcement action and the presence of at least one member of a finance-related committee who represents the state where the bank is headquartered. Moreover, we find that sub-committee chairmen have only a marginally significant influence (10%) over regulatory enforcement likelihood. In unreported regressions, we also explore sub-committee chairmanship and committee membership on the enforcement likelihood of less severe type of actions <sup>32</sup>.

[Insert Table 14 Around Here]

## 5.2 Variation across bank-related traits

In this section we aim to identify whether banks with particular profiles, may be more prone to preferential treatment. For this purpose, we explore certain traits of the bank's parent company. More precisely, we examine the ownership status, by considering whether the parent company is listed in a stock exchange or not. We also consider other characteristics, such as the parent's age, and we split the sample into banks with age above and below average.

Moreover, we take into account the diversity of geographic operation, and we distinguish between parent banks that have presence through offices or subsidiaries in multiple States and banks that operate in a single State <sup>33</sup>. We obtain the aforementioned information from

 $<sup>^{32}</sup>$ Further to the above, we also explore the potential impact of the scenario where there is both a chairman and member of the committee in a particular state. The results obtained do not provide any evidence on this matter. However, it should be noted that out of the 360,000 observations in the sample, there are only 12,857 cases were this variable takes the value of 1, and it takes the value of 0 in all other cases. Furthermore, out of the 12,857 cases, there are only 22 cases that correspond to an enforcement action. Thus, the insignificance could possibly be due to these particular characteristics of the sample, the results should be treated with caution, and this issue should be explored further. To conserve space, we do not tabulate the results, however they are available upon request

 $<sup>^{33}</sup>$ We do not split the sample on above/below average sub-samples, on the basis of the number of States where a bank operates, as this would result in having a considerable low number of enforcement actions in

SNL Financial database. Using the parent's unique regulatory id, we match this information with our main working sample. Due to coverage we drop institutions for which we do not have information. We re-estimate the baseline regression while splitting the sample, in turn, along these three dimensions. The results are presented in Table 15.

First, we observe that the Chair dummy, is statistically significant and is inversely related to enforcement action likelihood in the case of both listed and unlisted banks. Turning to the parent banks' age, the results show that the influence of the Chair is statistically significant at the 1% in the case where parent companies are above average age and 5% in the case of parent companies with below average age. There are at least two potential explanations for this finding. First, older banks may be more efficient and less likely to fail, influencing the decisions of regulators and politicians <sup>34</sup>. Second, older firms may be more experienced with the machinery of the state and political influence or they can be more valuable to governmental bodies (Desbordes and Vauday, 2007; Macher et al., 2011) <sup>35</sup>. Finally, the coefficient of the Chair dummy retains its negative sign in both the samples of multiple state and single state banks; however, it is statistically significant only in the latter. Therefore, it seems that only banks that operate locally benefit from preferential treatment. Nonetheless, this finding should possibly treated with some caution due to the very low number of enforcement actions in the case of the sample with banks operating in

the above average sub-sample. Therefore, we opt for a dummy variable that brings together all the banks operating in more than one State.

<sup>&</sup>lt;sup>34</sup>For example, DeYoung and Hasan (1998) report that profit efficiency improves rapidly at the typical de novo US bank during the first three years of operation; however, on average it takes about nine years to reach established bank levels. DeYoung (2003) examines banks chartered between 1980 and 1985 and concludes that initially, new banks were no more likely to fail than established banks; however, they become substantially more likely to fail as fast growth and negative earnings depleted their capital. In a more recent study, Lee and Yom (2016) also find that compared to small established US banks, recent de novos were financially fragile and failed at higher rates during the recent crisis.

<sup>&</sup>lt;sup>35</sup>For example, Desbordes and Vauday (2007) find that age has a positive impact on the political influence of foreign firms over government regulations. In another cross-country study, Macher et al. (2011) provide evidence that older firms are more effective in influencing governmental decision-making entities than younger firms in the cases of executive, legislative, and ministerial branches, albeit not in the case of regulatory agencies. Macher et al. (2011) provide three potential reasons for which age could matter, in general, in governmental outcomes. First, firms become more adept via "learning curve" effects in influencing governmental decision-makers. Second, firms that are unsuccessful in influencing governments may fail. Third, because older firms are more likely to survive, governmental decision-makers consider favourable decisions to older firms as more beneficial due to repeated and ongoing interactions.

multiple states.

[Insert Table 15 Around Here]

### 5.3 Social and political environment

In the previous sections, we explored factors related to committee and politician characteristics, as well as bank-related traits that could be useful in explaining the enforcement action likelihood. However, what we have not taken into account so far are institutional, political, and social attributes at the State level. These are important environmental characteristics that could shape both the level of pressure to the politicians and their reaction. Therefore, in the subsections that follow we consider, the State-level: (i) Economic freedom and corruption, (ii) Political orientation and (iii) Social norms.

#### 5.3.1 State economic freedom and corruption

We start by exploring whether a state's level of economic freedom and corruption are conditional factors driving the relationship between enforcement action likelihood and political influence. We opt to explore these elements, as they are key attributes to how well a democratic government functions. One would expect that in states with higher levels of economic freedom and lower levels of corruption, rent-seeking behavior would be less pronounced and elected officials would act in the interest of the citizens<sup>36</sup>.

In measuring economic freedom, we follow prior studies (e.g. Gropper et al., 2015) and obtain information on state Economic Freedom from 2000-2015 provided on an annual basis by the Fraser Institute  $^{37}$ . We make use of the state-level overall index of economic freedom

<sup>&</sup>lt;sup>36</sup>For example, Braun and Raddatz (2010) find that countries where banks are more politically connected are shown to rank higher on corruption. Similarly, Faccio et al. (2006) finds that corporate political connections are less common in the presence of more stringent regulations for political conflicts of interest and more common in countries that are highly corrupt. In addition, Campos and Giovannoni (2007) reveal that lobbying and corruption can be seen as substitutes, however, they suggest that lobbying as form of political influence is more effective than corruption in countries that are poorer and less developed. Finally, Damania et al. (2004)provide evidence suggesting that corruption is tied to lower levels of regulatory compliance.

<sup>&</sup>lt;sup>37</sup>Further details on the data can be found here:https://transition.fec.gov/pubrec/electionresults.shtml.

for North America, which consists of five main components/sub-groups for its calculation: Size of government, taxation, regulation, legal systems and property rights, sound money and freedom to trade internationally. The value of the index ranges from 1-10, where higher values reflect higher levels of economic freedom. Based on this information, we split the sample into high economic freedom and low economic freedom States, depending on whether the economic freedom of a given State is above or below average in a particular year.

To measure corruption on a state level we follow prior studies and collect information on convictions of public officials provided by the U.S Department of Justice <sup>38</sup> (e.g. Smith, 2016). Information is available on district level, therefore we aggregate figures on state level in order to match information with our sample. We then express these figures on per capital basis, and classify the states into ones with high corruption and low corruption, similarly to the case of economic freedom. We then re-estimate our baseline model for these sub-samples. The results are reported in Table 14.

### [Insert Table 16 Around Here]

Our findings show that while the effect of the Committee's Chair remains negative in both cases in regard to economic freedom (high and low). Turning to corruption, it is statistically significant only in States high corruption.

#### 5.3.2 Social norms

We now turn our focus on state-level social norms and their influence on the relationship between political influence and regulatory enforcement. In order to capture community morals and culture on a local level, we consider a state's level of religiosity. Prior studies have considered such proxies in both national (e.g. Dyreng et al., 2012; Adhikari and Agrawal, 2016) and international samples (e.g La Porta et al., 1999). Religion has been shown to have an effect on various aspects ranging from firm risk-taking (e.g. Adhikari and Agrawal, 2016; Hilary and Hui, 2009), misconduct (e.g. Grullon et al., 2009) or financial reporting (e.g. Dyreng

<sup>&</sup>lt;sup>38</sup>Full reports with relevant data https://www.justice.gov/criminal/pin.

et al., 2012). Moreover, in the context of politics and voting, empirical precedent shows that a community's level of religiousness can influence the voting behavior of individuals (e.g. Layman, 1997), as well as members of the US Congress (e.g. Fastnow et al., 1999). In the context of our study, religiosity could potentially influence the behavior of the politician in order to "serve" the beliefs of his or her potential voters. However, what is important to highlight is that this would not necessarily reflect the beliefs or ideas of the politician.

In order to test the above, we collect information from the "Churches and church membership files of American religion data archive  $(ARDA)^{"39}$ . In particular we retrieve data on religion that are made available from surveys conducted in 2000 and 2010. We follow Adhikari and Agrawal (2016) and obtain estimates for intermediate years by linearly interpolating the decennial data. We make use of state-level version of data for consistency with the set-up of our working sample. Our measure of religiosity is defined as the total number of congregations divided by a state's population. Based on this variable, we then split our sample to above ("*Religiosity high*") and below average religious ("*Religiosity low*") state and estimate our baseline model. Results are shown in Table 17.

Our key findings show that our political influence variable is negative and statistically significant for states with low levels of religiosity; whereas for states with high level of religiosity the effect is no longer significant. A potential explanation could be that in areas with higher religious adherence, the local community would follow religious beliefs and norms; while "punishing" misconduct and unethical decisions and behaviors. Thus, in such communities, politicians aiming to reassuring the probability of re-election and social acceptance, would be more likely to perhaps conform to such situations. In contrast, when religion is less pronounced or is not as popular within a local community, then the above "unwritten rules" would deteriorate. Consequently, in such cases there is further room for less ethical behavior, the cultivation of beneficial relationships between politicians and firms and other rent-seeking behavior.

 $<sup>^{39}{\</sup>rm More}$  information on the ARDA surveys can be found here http://www.thearda.com/Archive/ChState.asp.

#### [Insert Table 17 Around Here]

#### 5.3.3 State political polarization

Our final exercise is to examine the state level political polarization. In a two-party political system, as in the US, it is very likely to observe divergence of ideologies and political attitudes towards various socio-economic issues, within Congress among Republican and Democratic parties. Academic studies state that increased polarization impedes policy making and outcomes, and could also be tied to lower quality of legislation (e.g. Epstein and Graham, 2007). Moreover, it potentially drives Congress to "gridlock" and policy inaction (e.g. Binder, 2004; Jones, 2001). Our goal, in the context of the present study, is to explore whether ideological distance between political elites, enhances the observed relationship between political influence and favourable treatment of banks. In essence, it could be that in states with higher divergence of political attitudes by members of congress; politicians are more prone in establishing relationships with banks in order to secure their support, in exchange for benefits such as more favourable treatment.

We obtain information provided by Voteview on Congressional Roll-Call Votes  $^{40}$ , which provides scores of measures (DW-NOMINATE scores) that indicate on the ideological positions of elected officials over time and range from -1 (liberal) to 1 (conservative). We proceed to the construction of an "overall state polarization index", following the procedure in Mc-Carty et al. (1997), by calculating the average distance between Democrats and Republicans within a state, based on these scores. Once the index is constructed, we then create an indicator for states with high levels of elite polarization ("*Polarisation high*") and another for states with low levels of elite polarization ("*Polarisation low*"). Using these variables, we re-estimate our baseline model for the two samples. The results are reported in Table 15.

Our results suggest that the effect of our political influence dummy on enforcement action likelihood is significantly negative for states with higher levels of polarization, whereas

<sup>&</sup>lt;sup>40</sup>Further information can be found here https://voteview.com/data.

the coefficient of this variable is not statistically significant for states with lower levels of polarization. These findings could therefore suggest that indeed as members of congress become more diverge to each other, political influence is enhanced. Thus, it is consistent with prior literature suggesting that as polarization increases, policy outcomes are affected.

## 6 Conclusion

The core objective of the present study is to provide evidence on whether powerful politicians can exert influence on regulatory decision making. So far, the literature provides evidence that political connections are beneficial to firms, through a variety of channels. The majority of studies examining the relationship between political connections and regulatory outcomes, focus on the political strategies that firms adopt (e.g. lobbying, campaign contributions) to influence policy makers. Nonetheless, there is scarce evidence on whether and how political connections to Washington can result in beneficial regulatory treatment of banking instituions. The present study addresses this gap.

Focusing on the US banking sector, we use a sample of commercial banks for the period 2000-2015, and we measure political power with a binary variable that indicates whether or not a bank's parent company operates in a state where a Senator or Congressman serves as Chairman in a finance related Congressional Committee. We consider both severe and less severe enforcement actions issued against banks as a proxy for regulatory treatment, although we mainly focus on severe ones.

Our findings suggest an inverse relationship between political influence and enforcement action likelihood. Our results hold when alternative model specification and econometric methods are implemented. In a series of robustness tests, we also find that our political influence variable exerts significant influence on enforcement action likelihood when we control for the crisis years, for different regulatory agencies supervising the banks, as well as for several sub-samples (e.g. large banks, poor vs top capitalized banks). Our findings suggest that politicians with power in Congress are likely to have the ability to influence regulatory decision-making. Thus, these findings complement previous studies (e.g. Gropper et al., 2013,2015) that suggest that such connections have an impact on bank performance.

We further explore alternative patterns of influence on a politician, bank and state-level, in order to explain our baseline results. In particular, we find that Chairing a committee matters when Chairs are more senior, receive greater amounts of campaign contributions from the financial services industry or when they have higher levels of vote concentration. On a bank level, we find that political influence is important in the case of both listed and unlisted parent banks; however, the Chair is more influential in the case of older banks and banks operating in a single State. Finally, a state's overall socioeconomic and political environment also plays an important role in the treatment of banks with headquarters in politically important states. Overall, our findings are useful in shedding light on the impact of political influence on shaping regulatory decisions. Of course, there are still areas within this topic that remain unexplored due to data limitations and that could potentially be addressed in future research. For instance, future research could examine changes from less severe to severe enforcement actions and the opposite and their association with political connections, thus providing further insight on the value of political connections. Last but not least, our study adds value from a policy-making perspective, as we provide evidence that enables the further understanding of whether political influence enhances or impedes the effectiveness of banking supervision.

## References

- Acharya, V. V. and Yorulmazer, T. (2007). Too many to fail. an analysis of time-inconsistency in bank closure policies. *Journal of Financial Intermediation*, 16(1):1–31.
- Adams, R. (2013). Bankers on fed boards: Is good news for the banks bad news for the fed? Working Paper. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract\_ id=2361190.
- Adhikari, B. K. and Agrawal, A. (2016). Does local religiosity matter for bank risk-taking? Journal of Corporate Finance, 38:272–293.
- Agarwal, S., Lucca, D., Seru, A., and Trebbi, F. (2014). Inconsistent regulators: Evidence from banking. *The Quarterly Journal of Economics*, 129(2):889–938.
- Ai, C. and Norton, E. C. (2003). Interaction terms in logit and probit models. *Economics Letters*, 80(1):123–129.
- Austin, P. C. (2011). An introduction to propensity score methods for reducing the effects of confounding in observational studies. *Multivariate behavioral research*, 46(3):399–424.
- Barth, J. R., Caprio, G., and Levine, R. (2012). Guardians of finance: Making regulators work for us. Mit Press.
- Berry, C. R. and Fowler, A. (2017). Congressional committees, legislative influence, and the hegemony of chairs. *Journal of Public Economics*, 158:1–11.
- Bertrand, M., Kramarz, F., Schoar, A., and Thesmar, D. (2018). The cost of political connections. *Review of Finance*, 22(3):849–876.
- Besley, T. and Coate, S. (2001). Lobbying and welfare in a representative democracy. *The Review of Economic Studies*, 68(1):67–82.

- Binder, S. A. (2004). Stalemate: Causes and consequences of legislative gridlock. Brookings Institution Press.
- Blau, B. M. (2017). Lobbying, political connections and emergency lending by the federal reserve. *Public Choice*, 172(3-4):333–358.
- Borghesi, R. and Chang, K. (2015). The determinants of effective corporate lobbying. *Journal* of *Economics and Finance*, 39(3):606–624.
- Boubakri, N., Guedhami, O., Mishra, D., and Saffar, W. (2012). Political connections and the cost of equity capital. *Journal of corporate finance*, 18(3):541–559.
- Braun, M. and Raddatz, C. (2010). Banking on politics: When former high-ranking politicians become bank directors. *The World Bank Economic Review*, 24(2):234–279.
- Campos, N. F. and Giovannoni, F. (2007). Lobbying, corruption and political influence. *Public Choice*, 131(1-2):1–21.
- Chavaz, M. and Rose, A. K. (2016). Political borders and bank lending in post-crisis america. Technical report, National Bureau of Economic Research. Working Paper No. 22806.
- Chen, Y.-S., Shen, C.-H., and Lin, C.-Y. (2014). The benefits of political connection: Evidence from individual bank-loan contracts. *Journal of Financial Services Research*, 45(3):287–305.
- Claessens, S., Feijen, E., and Laeven, L. (2008). Political connections and preferential access to finance: The role of campaign contributions. *Journal of Financial Economics*, 88(3):554–580.
- Coates, D., Heckelman, J. C., and Wilson, B. (2011). Special-interest groups and growth. *Public Choice*, 147(3-4):439–457.
- Coates, D. and Wilson, B. (2007). Interest group activity and long-run stock market performance. *Public Choice*, 133(3-4):343–358.

- Cohen, L., Coval, J., and Malloy, C. (2011). Do powerful politicians cause corporate downsizing? *Journal of Political Economy*, 119(6):1015–1060.
- Cole, R. A. and White, L. J. (2012). Déjà vu all over again: The causes of US commercial bank failures this time around. *Journal of Financial Services Research*, 42(1-2):5–29.
- Cooper, M. J., Gulen, H., and Ovtchinnikov, A. V. (2010). Corporate political contributions and stock returns. *The Journal of Finance*, 65(2):687–724.
- Correia, M. M. (2014). Political connections and SEC enforcement. Journal of Accounting and Economics, 57(2):241–262.
- Damania, R., Fredriksson, P. G., and Mani, M. (2004). The persistence of corruption and regulatory compliance failures: theory and evidence. *Public choice*, 121(3-4):363–390.
- Delis, M. D., Staikouras, P. K., and Tsoumas, C. (2016). Formal enforcement actions and bank behavior. *Management Science*, 63(4):959–987.
- Deloitte (2015). Enforcement actions in the banking industry: Trends and lessons learned. Deloitte Center for Financial Services.
- Desbordes, R. and Vauday, J. (2007). The political influence of foreign firms in developing countries. *Economics & Politics*, 19(3):421–451.
- DeYoung, R. (2003). De novo bank exit. Journal of Money, Credit and Banking, 35(5):711–728.
- DeYoung, R. and Hasan, I. (1998). The performance of de novo commercial banks: A profit efficiency approach. *Journal of Banking & Finance*, 22(5):565–587.
- Disli, M., Schoors, K., and Meir, J. (2013). Political connections and depositor discipline. Journal of Financial Stability, 9(4):804–819.

- Duchin, R. and Sosyura, D. (2012). The politics of government investment. Journal of Financial Economics, 106(1):24–48.
- Duso, T. (2005). Lobbying and regulation in a political economy: evidence from the us cellular industry. *Public Choice*, 122(3-4):251–276.
- Dyreng, S. D., Mayew, W. J., and Williams, C. D. (2012). Religious social norms and corporate financial reporting. *Journal of Business Finance & Accounting*, 39(7-8):845– 875.
- Epstein, D. and Graham, J. D. (2007). Polarized politics and policy consequences. Rand Corporation.
- Faccio, M., Masulis, R. W., and McConnell, J. (2006). Political connections and corporate bailouts. *The Journal of Finance*, 61(6):2597–2635.
- Faccio, M. and Parsley, D. C. (2009). Sudden deaths: Taking stock of geographic ties. Journal of Financial and Quantitative Analysis, 44(3):683–718.
- Fan, J. P., Wong, T. J., and Zhang, T. (2007). Politically connected ceos, corporate governance, and post-ipo performance of china's newly partially privatized firms. *Journal of financial economics*, 84(2):330–357.
- Fastnow, C., Grant, J. T., and Rudolph, T. J. (1999). Holy roll calls: Religious tradition and voting behavior in the US house. *Social Science Quarterly*, 80:687–701.
- Firth, D. (1993). Bias reduction of maximum likelihood estimates. *Biometrika*, pages 27–38.
- Fisman, R. and Wang, Y. (2015). The mortality cost of political connections. The Review of Economic Studies, 82(4):1346–1382.
- Fulmer, S., Knill, A., and Yu, X. (2017). Negation of sanctions: the personal effect of political contributions. AFA San Diego Meetings paper. Available at https://papers.ssrn.com/ sol3/papers.cfm?abstract\_id=2024069#.

- Gibson, R., Odabasioglu, A., and Padovani, M. (2018). The determinants of banks' lobbying activities before and during the 2007-2009 financial crisis. Working Paper. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=1785435.
- Global Financial Centers Index Report (2018). Z/YEN. Available http://archive. longfinance.net/Publications/GFCI23.pdf.
- Goldman, E., Rocholl, J., and So, J. (2013). Politically connected boards of directors and the allocation of procurement contracts. *Review of Finance*, 17(5):1617–1648.
- Gropper, D. M., Jahera, J. S., and Park, J. C. (2013). Does it help to have friends in high places? Bank stock performance and congressional committee chairmanships. *Journal of Banking & Finance*, 37(6):1986–1999.
- Gropper, D. M., Jahera, J. S., and Park, J. C. (2015). Political power, economic freedom and congress: Effects on bank performance. *Journal of Banking & Finance*, 60:76–92.
- Grossman, G. M. and Helpman, E. (2001). Special interest politics. MIT press.
- Grullon, G., Kanatas, G., and Weston, J. (2009). Religion and corporate (mis) behavior. Working Paper, Rice University. Available at: https://papers.ssrn.com/sol3/papers. cfm?abstract\_id=1472118.
- Hilary, G. and Hui, K. W. (2009). Does religion matter in corporate decision making in america? *Journal of Financial Economics*, 93(3):455–473.
- Hill, M. D., Kelly, G. W., Lockhart, G. B., and Ness, R. A. (2013). Determinants and effects of corporate lobbying. *Financial Management*, 42(4):931–957.
- Holburn, G. L. and Bergh, R. G. V. (2008). Making friends in hostile environments: Political strategy in regulated industries. Academy of Management Review, 33(2):521–540.
- Houston, J. F., Jiang, L., Lin, C., and Ma, Y. (2014). Political connections and the cost of bank loans. *Journal of Accounting Research*, 52(1):193–243.

- Igan, D. and Lambert, T. (2018). Bank lobbying: Regulatory capture and beyond. Working Paper. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract\_id= 3128829.
- Igan, D. and Mishra, P. (2014). Wall street, capitol hill, and k street: Political influence and financial regulation. *The Journal of Law and Economics*, 57(4):1063–1084.
- Igan, D., Mishra, P., and Tressel, T. (2012). A fistful of dollars: lobbying and the financial crisis. NBER Macroeconomics Annual, 26(1):195–230.
- Infante, L. and Piazza, M. (2014). Political connections and preferential lending at local level: Some evidence from the italian credit market. *Journal of Corporate Finance*, 29:246–262.
- Ioannidou, V. P. (2005). Does monetary policy affect the central bank's role in bank supervision? Journal of Financial Intermediation, 14(1):58–85.
- Johnson, S. and Mitton, T. (2003). Cronyism and capital controls: evidence from malaysia. Journal of financial economics, 67(2):351–382.
- Jones, D. R. (2001). Party polarization and legislative gridlock. *Political Research Quarterly*, 54(1):125–141.
- Kedia, S. and Rajgopal, S. (2011). Do the SEC's enforcement preferences affect corporate misconduct? Journal of Accounting and Economics, 51(3):259–278.
- Kerr, W. R., Lincoln, W. F., and Mishra, P. (2014). The dynamics of firm lobbying. American Economic Journal: Economic Policy, 6(4):343–79.
- King, G. and Zeng, L. (1999). Estimating absolute, relative, and attributable risks in casecontrol studies. *Department of Government, Harvard University*.
- King, G. and Zeng, L. (2001). Logistic regression in rare events data. *Political analysis*, 9(2):137–163.

- Kostovetsky, L. (2015). Political capital and moral hazard. *Journal of Financial Economics*, 116(1):144–159.
- Kroszner, R. S. and Strahan, P. E. (1999). What drives deregulation? Economics and politics of the relaxation of bank branching restrictions. *The Quarterly Journal of Economics*, 114(4):1437–1467.
- La Porta, R., Lopez-de Silanes, F., Shleifer, A., and Vishny, R. (1999). The quality of government. The Journal of Law, Economics, and Organization, 15(1):222–279.
- Lambert, T. (2019). Lobbying on regulatory enforcement actions: Evidence from banking. Management Science. Forthcomming.
- Layman, G. C. (1997). Religion and political behavior in the United States: The impact of beliefs, affiliations, and commitment from 1980 to 1994. *Public Opinion Quarterly*, 61(2):288–316.
- Lee, Y. Y. and Yom, C. (2016). The entry, performance, and risk profile of de novo banks. FDIC Center for Financial Research, Working Paper No. 2016-03. Available at: https: //papers.ssrn.com/sol3/papers.cfm?abstract\_id=2778168.
- Lim, I., Hagendorff, J., and Armitage, S. (2016). Does distance impede regulatory monitoring? evidence from the banking industry.
- Lin, K. Z., Mills, L. F., Zhang, F., and Li, Y. (2018). Do political connections weaken tax enforcement effectiveness? *Contemporary Accounting Research*. Available at https://onlinelibrary.wiley.com/doi/abs/10.1111/1911-3846.12360 Accepted author manuscript.
- List, J. A., Millimet, D. L., Fredriksson, P. G., and McHone, W. W. (2003). Effects of environmental regulations on manufacturing plant births: evidence from a propensity score matching estimator. *Review of Economics and Statistics*, 85(4):944–952.

- Lucca, D., Seru, A., and Trebbi, F. (2014). The revolving door and worker flows in banking regulation. *Journal of Monetary Economics*, 65:17–32.
- Macher, J. T., Mayo, J. W., and Schiffer, M. (2011). The influence of firms on government. *The BE Journal of Economic Analysis & Policy*, 11(1).
- Mailath, G. J. and Mester, L. J. (1994). A positive analysis of bank closure. Journal of Financial Intermediation, 3(3):272–299.
- Mathur, I., Singh, M., Thompson, F., and Nejadmalayeri, A. (2013). Corporate governance and lobbying strategies. *Journal of Business Research*, 66(4):547–553.
- McCarty, N. M., Poole, K. T., and Rosenthal, H. (1997). Income redistribution and the realignment of american politics. Washington, DC: American Enterprise Institute Press.
- McLannahan, В. and Jopson, В. (2017).Wall Street spends record \$2bnUSelection lobbying. https://www.ft.com/content/ onRetrieved from 5060844a-0420-11e7-ace0-1ce02ef0def9.
- Mian, A., Sufi, A., and Trebbia, F. (2010). The political economy of the us mortgage default crisis. *The American Economic Review*, 100(5):1967–1998.
- Nys, E., Tarazi, A., and Trinugroho, I. (2015). Political connections, bank deposits, and formal deposit insurance. *Journal of Financial Stability*, 19:83–104.
- Papadimitri, P., Staikouras, P., Travlos, N. G., and Tsoumas, C. (2019a). Punished banks' acquisitions: Evidence from the us banking industry. *Journal of Corporate Finance*, 58:744– 764.
- Papadimitri, P., Wohlschlegel, A., et al. (2019b). Lobbying, regulatory enforcement and corporate governance: Theory and evidence from regulatory enforcement actions against us banks. Technical report, University of Portsmouth, Portsmouth Business School, Economics and Finance âĂę.

- Peltzman, S. (1976). Toward a more general theory of regulation. The Journal of Law and Economics, 19(2):211–240.
- Posner, R. A. (1974). Theories of economic regulation. The Bell Journal of Economics and Management Science, 5(2):335–358.
- Schiffer, M. and Weder, B. (2001). Firm size and the business environment: Worldwide survey results. International Finance Corporation, Washington, D.C., Discussion Paper No. 43.
- Shive, S. A. and Forster, M. M. (2016). The revolving door for financial regulators. *Review* of *Finance*, 21(4):1445–1484.
- Smith, J. D. (2016). US political corruption and firm financial policies. Journal of Financial Economics, 121(2):350–367.
- Stigler, G. J. (1971). The theory of economic regulation. The Bell journal of economics and management science, pages 3–21.
- Tomz, Michael, K. G. and Zeng, L. (1999). Relogit: Rare events logistic regression. Cambridge, MA: Harvard University.
- Vidal, J. B. I., Draca, M., and Fons-Rosen, C. (2012). Revolving door lobbyists. The American Economic Review, 102(7):3731–3748.
- Wu, W., Johan, S. A., and Rui, O. M. (2016). Institutional investors, political connections, and the incidence of regulatory enforcement against corporate fraud. *Journal of Business Ethics*, 134(4):709–726.
- Yu, F. and Yu, X. (2011). Corporate lobbying and fraud detection. Journal of Financial and Quantitative Analysis, 46(6):1865–1891.

## Table 1: List of variables

Variable	Description
A. Political connection variables	
	Takes,"1" if bank's parent company operates in a state where an elected official
Committee Chair	(Senator or Congressman)holds a chair position in a Congressional Committee
	that oversees the banking and financial services industry in a given year, 0 otherwise
	Takes "1" if bank's parent company operates in a state where an elected official
Committee member	(Senator or Congressman) holds a chair position in a Congressional Committee
	that oversees the banking and financial services industry in a given year, 0 otherwise
	Takes "1" if bank's parent company operates in a state where an elected official
Subcommittee chair	(Senator or Congressman) holds a chair position in a Congressional Committee
	that oversees the banking and financial services industry in a given year, 0 otherwise
Lobbying dummy	Takes,"1" if bank is engaged in lobbying in a given year and $0$ otherwise
B. Regulatory Variables	,
Action all types	Takes "1" if bank received any type of enforcement action in quarter t.0 otherwise
Severe action	Takes"1" if bank received severe type of enforcement action in quarter t, 0 otherwise
Less severe action	Takes "1" if bank received less severe type of enforcement action in quarter t, 0 otherwise
C. Financial and Demographic Characterist	ics of the second se
Risk-based capital ratio	Total qualifying capital divided by risk-weighted assets net of allowances and other reductions
Risk-weighted assets	Risk-weighted assets net of allowances and other reductions divided by total assets
Return on assets	Income (loss) before applicable income taxes and discontinued operations divided by total assets
Liquidity ratio	Cash and cash balances plus US treasury securities divided by total assets
Size	Natural logarithm of total assets
Listed (parent)	Takes "1" if bank's parent company is listed in a stock exchange, 0.0therwise
Age (parent)	Natural logarithm of number of years that the bank's parent company is operating
Multiple states (parent)	Number of states that parent company has offices
Distance to regulator	Distance in km between bank's headquarters and nearest Federal Reserve bank (Main office or branch)
D. Politician's characteristics, elections & v	oting information
	% total amount of contributions received by financial sector divided by the total amount
Contributions(fin) %	of contributions received
Votes %	%total number of votes received divided by the total number of votes within state or district
Seniority	Average combined age of chairmen in a given year
E. Macroeconomic and state level variables	
E. E.	State- level overall index of economic freedom that takes values 1-10. Higher values indicate
Economic Freedom	higher levels of economic freedom.
Corruption	Per capita number of convictions of public officials
Religiosity	Total number of congregations divided by a state's population
	Index reflecting the level of political polarization within a state. Calculated using
Political polarization	the average distance of DW scores for Democrats and Republicans.
-	Higher values indicate higher levels of political polarization.
GDP (state)	Natural logarithm of Gross domestic product per state
Population (state)	Natural logarithm of Population per state
Distance to regulator	Distance in km between a bank's headquarters and nearest Federal Reserve Bank (main office or branch)

Table 2: Descriptive statistics of enforcement action	$\mathbf{1S}$
---	---------------

Year	All types	Severe	Less Severe	Breakdown of se	evere actions			Severe	Action	s per regulatory agency
				Cease & Desist	Deposit ins.	Prompt Corr.	Formal written	FDIC	FED	OCC
					term.	Action	agreements	-		
2000	76	51	25	29	2	1	21	20	2	29
2001	111	60	51	31	0	3	26	25	2	33
2002	126	79	47	42	0	1	37	29	1	49
2003	143	70	73	51	0	0	22	37	0	33
2004	159	75	84	49	0	0	26	36	3	36
2005	171	65	106	48	0	0	22	30	0	35
2006	156	45	111	27	0	0	19	18	2	25
2007	185	63	122	48	0	0	16	39	1	23
2008	300	150	150	95	0	0	56	75	1	74
2009	593	364	229	232	0	27	106	243	11	110
2010	710	420	290	84	0	55	281	301	16	103
2011	450	228	222	163	1	37	27	169	5	54
2012	314	153	161	35	0	16	103	111	5	37
2013	186	72	114	21	0	8	45	47	0	25
2014	135	38	97	9	0	7	22	25	5	8
2015	109	22	87	5	0	2	15	18	2	2
Total	3.925	1955	1970	969	3	157	844	1223	56	676

Name	Year	Congress	Chamber	Committee	State	Party	Avg. service year in Congress	Aver. age while in chair service
William Philip Gramm	1999- 2000	106	Senate	Banking, Housing and Urban affairs	Texas	Republican	20.5	57.5
Paul Spyros Sarbanes	2001- 2002	107	Senate	Banking, Housing and Urban affairs	Maryland	Democratic	30.5	68.5
Richard Craig Shelby	2003- 2006	108- 109	Senate	Banking, Housing and Urban affairs	Alabama	Republican	25.5	70.5
Christopher John Dodd	2007- 2010	110- 111	Senate	Banking, Housing and Urban affairs	Connecticut	Democratic	33.5	64.5
Tim Johnson	2011- 2014	112- 113	Senate	Banking, Housing and Urban affairs	South Dakota	Democratic	26	67
Richard Craig Shelby	2015- 2016	114	Senate	Banking, Housing and Urban affairs	Alabama	Republican	37	82
Michael Garver Oxley	2001- 2006	107- 109	House	Financial Services	Ohio	Republican	30.5	59.5
Barney Frank	2007- 2010	110- 111	House	Financial Services	Massachusetts	Democratic	35.5	68.5
Spencer Bachus	2011- 2013	112- 113	House	Financial Services	Alabama	Republican	19	65
Jeb Hensarling	2013-	114-	House	Financial Services	Texas	Republican	11	57

Table 3: Summary information of banking committee chairs

	Observations	Mean	St. Dev.	Min	Max
Panel A: Financial & Po	litical influence	variables			
(i) Full sample					
Severe Actions	374528	0.0052333	0.0721518	0	1
Chair dummy	374528	0.0494703	0.2168481	0	1
Risk-based capital ratio	372851	0.1627947	0.0666491	-0.1676752	0.8203545
Risk weighted ratio	372124	0.6855651	0.1273082	0.2852844	1.081971
Return on Assets	374368	0.0069594	0.0096871	-0.1567464	0.1713443
Liquidity	369494	0.0594606	0.0514141	-0.0024057	0.3194084
Size	374528	11.94797	1.333464	3.332205	21.46335
(ii) Connected sample					
Risk-based capital ratio	18454	0.1753796	0.0707974	-0.001486	0.8168582
Risk weighted ratio	18381	0.6438928	0.1287669	0.285319	1.077301
Return on Assets	18516	0.0076363	0.0073914	-0.1132164	0.1699827
Liquidity	18120	0.0667662	0.058359	0.0026748	0.3192939
Size	18528	12.09121	1.411821	6.900731	19.43696
(iii) Punished sample					
Risk-based capital ratio	1952	0.1242376	0.0578782	-0.1469169	0.5605881
Risk weighted ratio	1945	0.7210423	0.1076446	0.2898475	1.075673
Return on Assets	1950	-0.0073306	0.022274	-0.1473495	0.115203
Liquidity	1926	0.0769846	0.0597405	0.0005808	0.3169498
Size	1955	12.32602	1.670912	8.840435	21.402
(iv) Punished and conne	cted sample				
Risk-based capital ratio	47	0.1380653	0.0394201	0.0442328	0.2327435
Risk weighted ratio	46	0.6758194	0.1176503	0.4384539	0.93757583
Return on Assets	47	-0.0003394	0.0148498	-0.056174	0.015958
Liquidity	47	0.0654366	0.0501277	0.0156473	0.2918928
Size	47	12.45925	1.858892	10.42736	18.28799
Panel B: Macroeconomic	and other state	e-level variab	oles		
Economic freedom	374439	6.99164	0.5510168	5.2	8.55
Corruption	374439	26.64706	26.91394	0	166
Religiosity	371360	0.0013299	0.0004546	0.0004646	0.0024771
Political Polarization	372897	0.7290545	0.1475768	0.228	1.102857
GDP (state)	371360	12.44682	0.9484076	9.761291	14.73414
Population (state)	371360	15.60248	0.878587	13.1109	17.4799
Distance to regulator	362030	571.6422	1800.932	0	8462.514

 Table 4: Descriptive statistics

Table 4 reports the descriptive statistics of the variables included in the analysis. For a detailed definition of variables see Table 1.The sample period is 2000q1 - 2015q. Panel A includes financial and political influence related information. Section i provides the descriptive statistics of the full sample. Sections ii and iii, provide the descriptive statistics for politically connected and punished with severe type enforcement actions banks, respectively. Panel B includes macroeconomic and other state-level variables used in the analysis.

Table 5: Univariate test of means for financial variables

	Mean	Mean	Difference
	(Punished with severe type action)	(Non-Punished)	(P-Value)
Political influence	0.0245524	0.049601	0.0000
Risk-based capital ratio	0.1242376	0.1629976	0.0000
Risk-weighted assets	0.7210423	0.6853787	0.0000
Return on assets	-0.0073306	0.0070343	0.0000
Liquidity	0.0769846	0.0593688	0.0000
Size	12.32602	11.94599	0.0000

Table 5 reports the mean values of each variable for two distinct groups; banks that received a severe type regulatory enforcement action (punished) and banks that did not (non-punished). The mean difference along with the p-value of the relevant statistical test is reported in the last column. For full definition of variables see Table 1.

	(1	)	(2	)		(3)
	All ac	tions	Severe	only		Less severe only
	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx
Committee chair	-0.107 ***	-0.002 ***	-0.178 ***	-0.002 ***	-0.052	-0.001
	(0.038)		(0.058)		(0.045)	
Risk-weighted assets ratio	0.088	0.002	-0.048	-0.000	0.217 **	0.003 **
	(0.076)		(0.099)		(0.091)	
Risk-based capital ratio	-2.185 ***	-0.045 ***	-3.288 ***	-0.030 ***	-0.884 ***	-0.011 ***
	(0.207)		(0.303)		(0.223)	
Return on assets	-19.002 ***	-0.393 ***	-20.555 ***	-0.188 ***	-10.260 ***	-0.122 ***
	(0.532)		(0.582)		(0.644)	
Liquidity ratio	0.861 ***	0.018 ***	1.087 ***	0.010 ***	0.492 ***	0.006 ***
	(0.132)		(0.167)		(0.173)	
Size	0.110 ***	0.002 ***	0.042 ***	0.000 ***	0.145 ***	0.002 ***
	(0.009)		(0.010)		(0.012)	
Constant	-3.378 ***		-2.511 ***		-4.467 ***	
	(0.172)		(0.210)		(0.217)	
Observations	366233		365600		366233	
Cluster	8698.000		8681.000		8698.000	
Pseudo $R^2$	0.083		0.125		0.054	
Time dummies	YES		YES		YES	
States dummies	YES		YES		YES	
Bank FE	NO		NO		NO	

## Table 6: Baseline results (Probit model specification)

Table 6 reports the baseline results of a Probit model. We report estimated coefficients and marginal effects at means. Estimations include robust standard errors clustered by bank. Sample period is 2000ql-2015q4. All financial control variables are lagged by one quarter. In Column 1, the dependent variable refers to all types of enforcement actions. Column 2 refers to severe type of actions only. Column 3 refers to less severe actions only. A set of full time and state dummies are included. For detailed variable description see Table 1. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

	(1)	(2)	(3)
	All actions	Severe only	Less Severe only
	eta / SE	eta / SE	eta / SE
Committee Chair	-0.002**	-0.002**	-0.001
	(0.001)	(0.001)	(0.001)
Risk-weighted assets ratio	-0.009***	-0.011***	0.002
	(0.003)	(0.002)	(0.002)
Risk-based capital ratio	-0.068***	-0.068***	-0.000
	(0.005)	(0.004)	(0.004)
Return on assets	-1.122***	-0.868***	-0.255***
	(0.023)	(0.016)	(0.017)
Liquidity ratio	$0.023^{***}$	$0.013^{***}$	$0.010^{***}$
	(0.004)	(0.003)	(0.003)
Size	$0.006^{***}$	0.003***	$0.004^{***}$
	(0.000)	(0.000)	(0.000)
Constant	-0.022	0.002	-0.024
	(0.039)	(0.027)	(0.028)
Observations	367019	367019	367019
Cluster	8723.000	8723.000	8723.000
$R^2$	0.0090	0.0106	0.0106
Time dummies	YES	YES	YES
State dummies	YES	YES	YES
Bank FE	YES	YES	YES

Table 7: Alternative model specification - Linear Probability Model

Table 7 reports the baseline results of a Linear Probability model. Estimations include robust standard errors clustered by bank. Sample period is 2000q1-2015q4. All financial control variables are lagged by one quarter. In Column 1, the dependent variable refers to all types of enforcement actions. Column 2 refers to severe type of actions only. Column 3 refers to less severe actions only. A set of full time and state dummies are included. Bank fixed effects are also introduced in this model. For detailed variable description see Table 1. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

	(1)	(2)	(3)
	All actions	Severe only	Less severe only
	$eta \;/\; { m SE}$	eta / SE	$eta \;/\; { m SE}$
Committee Chair	-0.342 ***	-0.658 ***	-0.144
	(0.106)	(0.171)	(0.129)
Risk-weighted assets ratio	0.537 ***	0.479 *	0.700 ***
	(0.187)	(0.270)	(0.242)
Risk-based capital ratio	-5.952 ***	-9.270 ***	-2.580 ***
	(0.555)	(0.833)	(0.633)
Return on assets	-38.370 ***	-42.282 ***	-23.237 ***
	(1.165)	(1.388)	(1.314)
Liquidity ratio	2.611 ***	3.986 ***	0.974 **
	(0.322)	(0.429)	(0.483)
Size	0.273 ***	0.148 ***	0.356 ***
	(0.019)	(0.024)	(0.029)
Constant	-7.382 ***	-6.157 ***	-9.662 ***
	(0.336)	(0.418)	(0.470)
Observations	367019	367019	367019
Pseudo $R^2$			
Time dummies	NO	NO	NO
State dummies	NO	NO	NO

Table 8: Alternative model specification - Rare events logistic model

Table 8 reports the baseline results when estimated using a rare events logistic regression model. Estimations include robust standard errors clustered by bank. Sample period is 2000q1-2015q4. All financial control variables are lagged by one quarter. In Column 1, the dependent variable refers to all types of enforcement actions. Column 2 refers to severe type of actions only. Column 3 refers to less severe actions only. A set of full time and state dummies are included. For detailed variable description see Table 1. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

	(1)	(2)	(3)
	All actions	Severe only	Less severe only
	eta / SE	eta / SE	$eta \;/\; { m SE}$
Committee Chair	-0.342***	-0.658***	-0.144
	(0.091)	(0.155)	(0.112)
Risk-weighted assets ratio	$0.537^{***}$	$0.479^{*}$	0.700***
	(0.169)	(0.249)	(0.227)
Risk-based capital ratio	-5.952***	-9.270***	-2.580***
	(0.438)	(0.646)	(0.569)
Return on assets	-38.370***	-42.282***	-23.237***
	(0.943)	(1.126)	(1.496)
Liquidity ratio	2.611***	3.986***	0.974**
	(0.309)	(0.414)	(0.463)
Size	0.273***	0.148***	0.356***
	(0.010)	(0.016)	(0.012)
Constant	-7.382***	-6.157***	-9.662***
	(0.204)	(0.308)	(0.267)
Observations	367019	367019	367019
Pseudo $R^2$			
Time dummies	NO	NO	NO
State dummies	NO	NO	NO

Table 9: Alternative model specification - Firth's PMLE model

Table 9 reports the baseline results when estimated using Firth's PMLE model. Estimations include robust standard errors clustered by bank. Sample period is 2000q1-2015q4. All financial control variables are lagged by one quarter. In Column 1, the dependent variable refers to all types of enforcement actions. Column 2 refers to severe type of actions only. Column 3 refers to less severe actions only. A set of full time and state dummies are included. For detailed variable description see Table 1. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

	(1)		3)	2)		3)		(4)	()	5)		(9)
	Financial cris	is years only	Excl. financia	al crisis years	Excl. low cap	italised banks	Excl. high cal	pitalised banks	Excl. large b	anks by state	Excl. large	banks by year
	$\beta$ / SE	Mfx	$\beta / SE$	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta / SE$	Mfx	$\beta$ / SE	Mfx
Comittee chair	-0.733**	-0.012**	-0.173**	-0.001**	-0.185**	-0.002**	-0.178**	-0.002**	$-0.159^{**}$	-0.001**	-0.166**	-0.001**
	(0.373)		(0.082)		(0.087)		(0.081)		(0.078)		(0.083)	
Risk-weighted assets	-0.355	-0.006	-0.117	-0.001	0.024	0.000	-0.067	-0.001	-0.024	-0.000	-0.033	-0.00
	(0.275)		(0.141)		(0.145)		(0.147)		(0.145)		(0.146)	
Risk-based capital ratio	$-5.376^{***}$	-0.088***	$-2.961^{***}$	$-0.024^{***}$	$-2.219^{***}$	$-0.020^{***}$	$-3.641^{***}$	$-0.033^{***}$	-3.382***	$-0.030^{***}$	$-3.446^{***}$	$-0.031^{***}$
	(0.970)		(0.428)		(0.351)		(0.429)		(0.412)		(0.417)	
Return on assets	$-17.108^{**}$	$-0.280^{***}$	$-20.568^{***}$	$-0.164^{***}$	$-19.541^{***}$	$-0.172^{***}$	$-20.667^{***}$	$-0.189^{***}$	$-20.659^{***}$	$-0.185^{***}$	$-20.706^{**}$	$-0.184^{***}$
	(1.664)		(0.720)		(0.886)		(0.783)		(0.804)		(0.801)	
Liquidity ratio	$1.456^{***}$	$0.024^{***}$	$1.158^{***}$	$0.009^{***}$	$0.903^{***}$	$0.008^{***}$	$1.071^{***}$	$0.010^{***}$	$1.026^{***}$	$0.009^{***}$	$1.016^{***}$	$0.009^{***}$
	(0.319)		(0.217)		(0.232)		(0.210)		(0.215)		(0.218)	
Size	0.023	0.000	$0.044^{***}$	$0.000^{***}$	$0.046^{***}$	$0.000^{***}$	$0.040^{***}$	$0.000^{***}$	$0.026^{**}$	$0.000^{**}$	$0.017^{*}$	0.000*
	(0.020)		(0.015)		(0.014)		(0.012)		(0.011)		(0.010)	
Constant	$-1.551^{***}$		$-2.609^{***}$		-2.721***		$-2.417^{***}$		$-2.314^{***}$		$-2.196^{***}$	
	(0.392)		(0.214)		(0.188)		(0.186)		(0.167)		(0.165)	
Observations	45827		319227		361917		362519		361967		362143	
Cluster	43.000		46.000		46.000		46.000		46.000		46.000	
Pseudo $R^2$	0.143		0.117		0.080		0.127		0.127		0.128	
Time dummies	YES		YES		YES		YES		YES		YES	
State dumnies	YES		YES		YES		$\mathbf{YES}$		YES		YES	
Table 10 reports the results of baseline model for years 2007 respectively. Columns 6 and unless otherwise stated. All f and 10% level.	of a set of additic 7-2011 and 2007-2 7 report the resu înancial control v	mal sensitivity t 2008 only respec ilts when excluc arriables are lag	ests. For each m trively. Column 3 ling large banks j ged by one quart	odel we report e s reports the resu per state and per er. A set of full t	stimated coefficie lts obtained whe r year respectivel ime and state du	nts and marginal n estimating the t y. Models are esti immies are include	effects at means. aseline model ext imated with a Pr 3d. For full varial	Results reported cluding the years 2 obit model with rr ble description see	in Columns 1 and 007-2008. In Col bbust standard ei Table 1. The *,	1 2 and refer to th lumns 4 and 5 we rrors clustered by ** and *** signs	te results obtained exclude high and ] bank. Sample per denote statistical s	when estimating the ow capitalised banks iod is $2000q1-2015q4$ ignificance at the 1,5

(H
vsis (
anal
tness
pus
$\mathbf{R}_{0}$
10:
Table

		(1	(2		))		(4)		(5)	
	Controlling	for lobbying	$\operatorname{Regulatory}$	agencies	Distance to	regulator	Excluding i banking	important states	Clusteri state l	ing at evel
	$eta \ / \ { m SE}$	Mfx	$eta \ / \ { m SE}$	Mfx	$eta \ / \ { m SE}$	Mfx	$\beta / SE$	Mfx	$\beta \mid SE$	Mfx
Committee chair	$-0.176^{**}$ (0.081)	-0.002**	$-0.173^{**}$ (0.080)	-0.001**	$-0.180^{**}$ (0.075)	-0.002**	$-0.178^{**}$ (0.082)	-0.002**	$-0.178^{**}$ (0.082)	-0.002**
Lobbying	0.124 (0.107)	0.001	~		~				~	
Risk-weighted assets ratio	-0.051	-0.000	0.024	0.000	-0.049	-0.000	-0.048	-0.000	-0.048	-0.000
Risk-based capital ratio	(0.143) -3.320***	-0.030***	(0.136) -3.508***	-0.027***	(0.133) -3.439***	$-0.031^{***}$	(0.143) -3.288***	-0.030***	(0.143) -3.288***	-0.030***
Return on assets	(0.419) -20.534***	-0.187***	(0.434)-20.718***	-0.161***	(0.422) -20.395***	-0.184***	(0.421) -20.555***	-0.188***	(0.421) -20.555***	-0.188***
Liquidity ratio	(0.780) 1.075***	$0.010^{***}$	(0.805) 1.168***	***600.0	(0.787) 1.063***	$0.010^{***}$	(0.784) 1.087***	$0.010^{***}$	(0.784) 1.087***	$0.010^{**}$
Size	(0.210) $0.037^{***}$	0.000***	(0.197) $0.038^{***}$	0.000***	(0.216) $0.047^{***}$	0.000***	(0.208) $0.042^{***}$	0.000***	(0.208) $0.042^{***}$	0.000***
	(0.010)		(0.012)	+ + - 	(0.013)		(0.012)		(0.012)	
FDIC			$0.585^{***}$ (0.056)	0.005***						
000			$0.800^{***}$ (0.046)	0.006***						
Distance to regulator					-0.000)	-0.000				
Constant	$-2.445^{***}$		$-3.002^{***}$		$-2.526^{***}$		$-2.511^{***}$		-2.511*** (0.186)	
Observations	365600		365600		353304		348857		365600	
Cluster	46.000		46.000		45.000		46.000		46.000	
Pseudo $R^2$	0.125		0.142		0.120		0.125		0.125	
Time dummies	YES		YES		YES		YES		YES	
State dummies	YES		YES		YES		YES		YES	
Table 11 reports the results of reported in Column 1 refer to t regulatory agency dummies. Co 5 reports the results obtained w A set of full time and state dun level.	a set of additio he results obtain blumns 3 and 4 r hen clustering s mmies are inclu	ual sensitivity to aed when contro eport the results tandard errors a ded. For full va	ests. For all me dling for an alte s obtained when t the state leve riable descripti	odel specificat ernative chanr a excluding im I. Sample peri on see Table 1	ions we report tel of political uportant bankii od is 2000q1-2 I. The *, ** a	estimated coe influence. Colu ug states and c 315q4. All fina nd *** signs d	fficients and m mns 2 reports - lustering at the ncial control va enote statistica	arginal effects the results obt e state level, re riables are lag l significance	at means. Rescaled when add spectively. Coll ged by one quan at the 1,5 and	ults ling ter. 10%

Table 11: Robustness analysis (2)

Table 12: Propensity score matching

	(1)	(2)
Estimator	ATT	Number of Matches
Nearest neighbor $(n=1; caliper=0.01)$	-0.0012 *	17,930
	(0.00072)	
Nearest neighbor $(n=10; caliper=0.01)$	-0.002 ***	17,930
	(0.00051)	
Nearest neighbor $(n=50; caliper=0.01)$	-0.0019 ***	17,930
	(0.00049)	
Nearest neighbor $(n=100; caliper=0.01)$	-0.0019***	17,930
	(0.00049)	

Table 12 reports the results from the Propensity Score matching. Column 1 reports the ATT estimates, which reveal the mean difference between the probability of a severe action among the treated and untreated group. Column 2 reports the total number of matches. Estimations make use of Near Neighbor for n=1,10,50 and 100 with caliper being equal to 0.01. Sample period is 2001q1-2015q4. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

	(1		(2)		(3		(4		(2			(9)
	High se	niority	Low ser	iiority	Financial se	ctor contr.	Financial se	ctor contr.	% Votes	received	% Vot	es received
	)	,		,	(above a	verage)	(below a	verage)	(Above a	average)	(Belov	v average)
	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$eta \ / \ { m SE}$	Mfx	$eta \mid { m SE}$	Mfx	$\beta$ / SE	Mffx
Committee chair	-0.216 ***	-0.002 ***	-0.121	-0.001	-0.235 ***	-0.003 ***	-0.065	-0.000	-0.170 ***	-0.002 ***	0.095	0.001
	(0.071)		(0.107)		(0.074)		(0.089)		(0.061)		(0.139)	
Risk-weighted assets ratio	-0.135	-0.001	-0.083	-0.000	-0.129	-0.001	-0.242	-0.001	-0.062	-0.001	0.031	0.000
	(0.113)		(0.223)		(0.123)		(0.180)		(0.109)		(0.251)	
Risk-based capital ratio	-3.781 ***	-0.040 ***	-2.354 ***	-0.012 ***	-4.286 ***	-0.049 ***	-2.117 ***	-0.013 ***	-4.073 ***	-0.039 ***	-0.508	-0.003
	(0.391)		(0.527)		(0.450)		(0.429)		(0.391)		(0.381)	
Return on assets	-20.572 ***	-0.218 ***	-16.441 ***	-0.084 ***	-19.871 ***	-0.226 ***	-16.793 ***	-0.102 ***	-20.973 ***	-0.201 ***	-12.063 ***	-0.075 ***
	(0.676)		(1.244)		(0.725)		(1.100)		(0.667)		(1.347)	
Liquidity ratio	1.372 ***	0.015 ***	0.141	0.001	1.356 ***	0.015 ***	0.140	0.001	1.012 ***	0.010 * * *	-0.424	-0.003
	(0.188)		(0.387)		(0.194)		(0.343)		(0.179)		(0.690)	
Size	0.040 ***	0.000 ***	0.039 *	0.000 *	0.040 ***	0.000 ***	0.022	0.000	0.044 ***	0.000 ***	0.003	0.000
	(0.010)		(0.022)		(0.011)		(0.020)		(0.011)		(0.027)	
Constant	-2.358 ***		-2.552 ***		-2.282 ***		-2.301 ***		-2.386 ***		-2.473 ***	
	(0.226)		(0.417)		(0.250)		(0.365)		(0.240)		(0.462)	
Observations	247990		110629		198531		158719		288281		66071	
Cluster	7582.000		7821.000		7136.000		7971.000		8126.000		6942.000	
Pseudo $R^2$	0.127		0.082		0.135		0.065		0.131		0.031	
Time dummies	ON		NO		NO		NO		NO		ON	
State dummies	$\mathbf{YES}$		$\mathbf{YES}$		$\mathbf{YES}$		YES		$\mathbf{YES}$		YES	
Table 13 reports the estimation seniority, contributions and voto set of state dummies are include	results when sen ss concentration od. For detailed	uiority, financial . We make use variable descrip	sector contribut of a Probit moc vion see Table 1	ions and % vo lel with robust . The *, ** an	tes received by e standard errors ad *** signs den	lected officials : s clustered by 1 ote statistical s	are taken into co ank. Sample po ignificance at th	msideration. Were a solution of the second s	e run the regres 2015q4. All fin level.	ssions separately ancial control v	r for sub-samples of ariables are lagged	periods with high/low by one quarter. A full

influence
and
power
Politician's
Table 13:

	(1	.)		(2)
Dep. Var: Severe type of e	nforcement a	ctions		
	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx
Committee member	-0.063***	-0.001***		
	(0.024)			
Sub-committee chairman			-0.124*	-0.001*
			(0.071)	
Risk-weighted assets ratio	0.000	0.000	-0.004	-0.000
	(0.100)		(0.100)	
Risk-based capital ratio	-3.252***	-0.029***	-3.259***	-0.029***
	(0.307)		(0.307)	
Return on assets	-20.345***	-0.184***	-20.331***	-0.184***
	(0.588)		(0.589)	
Liquidity ratio	$1.074^{***}$	$0.010^{***}$	$1.084^{***}$	$0.010^{***}$
	(0.170)		(0.170)	
Size	$0.043^{***}$	$0.000^{***}$	$0.044^{***}$	$0.000^{***}$
	(0.010)		(0.010)	
Constant	-2.497***		-2.538***	
	(0.213)		(0.213)	
Observations	362574		362574	
Cluster	8575.000		8575.000	
Pseudo $R^2$	0.118		0.118	
Time dummies	YES		YES	
State dummies	YES		YES	

Table 14: Committee membership and sub-committee chairmanship

Table 14 reports the estimation results when committee membership and sub-committee chairmanship are explored. We make use of a Probit model with robust standard errors clustered by bank. For all model specifications we report estimated coefficients and marginal effects at means. Sample period is 2000q1-2015q4. All financial control variables are lagged by one quarter. A full set of state dummies are included. For detailed variable description see Table 1. The \*, \*\* and \*\*\* signs denote statistical significance at the 1,5 and 10% level.

traits
Bank-related
15:
Table

			(2)		(3		(4		(5		9)	
	List	ed	Non-l	isted	Above a age p	average arent	Below a age pe	verage arent	Multiple	e state	Single	state
	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx	$\beta$ / SE	Mfx
Committee chair	-0.508**	-0.004**	-0.257***	-0.002***	$-0.405^{***}$	$-0.002^{***}$	-0.230**	-0.002**	-0.187	-0.001	-0.337***	-0.002***
	(0.223)		(0.094)		(0.144)		(0.112)		(0.156)		(0.108)	
Risk-weighted assets ratio	0.239	0.002	0.142	0.001	0.243	0.001	-0.018	-0.000	0.215	0.001	0.055	0.000
	(0.300)		(0.162)		(0.205)		(0.202)		(0.256)		(0.168)	
Risk-based capital ratio	-1.312*	$-0.010^{*}$	$-1.912^{***}$	$-0.012^{***}$	-1.288**	-0.007**	-2.205***	$-0.016^{***}$	0.065	0.000	-2.232***	$-0.014^{***}$
	(0.686)		(0.396)		(0.528)		(0.498)		(0.446)		(0.447)	
Return on assets	$-17.677^{***}$	$-0.128^{***}$	$-22.756^{***}$	$-0.138^{***}$	$-23.139^{***}$	$-0.121^{***}$	$-19.355^{***}$	$-0.141^{***}$	$-16.684^{***}$	$-0.101^{***}$	-23.023***	$-0.142^{***}$
	(1.700)		(1.071)		(1.639)		(1.063)		(1.840)		(1.116)	
Liquidity ratio	$2.562^{***}$	$0.019^{***}$	0.434	0.003	$1.378^{***}$	$0.007^{***}$	0.550	0.004	$2.487^{***}$	$0.015^{***}$	$0.462^{*}$	$0.003^{*}$
	(0.454)		(0.276)		(0.325)		(0.337)		(0.442)		(0.278)	
Size	$0.107^{***}$	$0.001^{***}$	0.024	0.000	$0.101^{***}$	$0.001^{***}$	-0.016	-0.000	$0.094^{***}$	$0.001^{***}$	$0.038^{**}$	$0.000^{**}$
	(0.016)		(0.018)		(0.015)		(0.022)		(0.018)		(0.017)	
Constant	-3.905***	* * *	-2.724***	* *	-3.828***	***	$-2.201^{***}$	* *	$-4.209^{***}$	* *	-2.724***	* * *
	(0.385)		(0.316)		(0.348)		(0.413)		(0.457)		(0.326)	
Observations	44095		195864		125131		117520		47648		194252	
Cluster	1348.000		4017.000		3619.000		4044.000		1556.000		3897.000	
Pseudo $R^2$	0.117		0.093		0.1111		0.087		0.111		0.098	
Time dummies	$\mathbf{YES}$		$\mathbf{YES}$		$\mathbf{YES}$		YES		$\mathbf{YES}$		YES	
State dummies	YES		$\mathbf{YES}$		YES		YES		YES		YES	
Table 15 reports the estimation we report estimated coefficients included. For detailed variable d	results when ba and marginal e lescription see 7	ank-related ch ffects at mean Table 1. The '	aracteristics ar. s. Sample perid *, ** and *** si	e explored. We od is 2000q1-20 gns denote sta	e make use of a 015q4. All fina tistical signific	a Probit mode uncial control v cance at the 1,	l with robust st ariables are lag 5 and 10% leve	andard errors ged by one qu l	clustered by b arter. A full se	ank. For all m t of state and	iodel specificati time dummies	ons are

	-/		()		/ •			
	r)	(	7			()		(4)
	Economic	: freedom	Economic	freedom	Corru	Iption	Corr	uption
	(Above a)	average)	(Below $\varepsilon$	werage)	(Above a)	average)	(Below	average)
	eta / SE	Mfx	$eta \ / \ { m SE}$	Mfx	eta / SE	Mfx	$eta \ / \ { m SE}$	Mfx
Committee chair	$-0.148^{**}$	-0.001**	-0.183**	-0.002**	-0.226***	-0.002***	-0.104	-0.001
	(0.072)		(0.088)		(0.070)		(0.098)	
Risk-weighted assets ratio	-0.119	-0.001	-0.006	-0.000	0.104	0.001	-0.147	-0.001
	(0.135)		(0.143)		(0.157)		(0.129)	
Risk-based capital ratio	$-2.658^{***}$	$-0.019^{***}$	$-4.094^{***}$	-0.043***	$-3.391^{***}$	$-0.031^{***}$	-3.428***	-0.029***
	(0.405)		(0.501)		(0.552)		(0.400)	
Return on assets	$-18.819^{***}$	$-0.137^{***}$	$-20.296^{***}$	$-0.213^{***}$	$-20.147^{***}$	$-0.185^{***}$	$-19.981^{***}$	$-0.168^{***}$
	(0.855)		(0.825)		(0.907)		(0.772)	
Liquidity ratio	$0.696^{***}$	$0.005^{***}$	$0.880^{***}$	$0.009^{***}$	$0.967^{***}$	$0.009^{***}$	$0.838^{***}$	$0.007^{***}$
	(0.264)		(0.238)		(0.273)		(0.226)	
Size	$0.033^{**}$	$0.000^{**}$	$0.034^{***}$	$0.000^{***}$	0.016	0.000	$0.045^{***}$	$0.000^{***}$
	(0.015)		(0.013)		(0.015)		(0.013)	
Population	0.029	0.000	$-0.495^{***}$	-0.005***	-0.488***	-0.004***	-0.234**	$-0.002^{**}$
	(0.115)		(0.111)		(0.131)		(0.098)	
GDP	0.027	0.000	$0.466^{***}$	$0.005^{***}$	$0.494^{***}$	$0.005^{***}$	$0.247^{***}$	$0.002^{***}$
	(0.103)		(0.093)		(0.111)		(0.084)	
Constant	-3.082***		-0.485		-1.227		$-1.793^{***}$	
	(0.888)		(0.749)		(0.936)		(0.679)	
Observations	182158		178170		145859		215940	
Cluster	5654.000		5696.000		6481.000		7758.000	
Pseudo $R^2$	0.102		0.132		0.110		0.132	
Times dummies	YES		YES		YES		YES	
State dumnies	YES		YES		YES		YES	
Table 16 reports the estimation for sub-samples of states with h bank. For all model specification variables are lagged by one quan-	results when st. iigh/low econoi is we report est rter. A full set	ate economic fi nic freedom an imated coeffici of time dumm	eedom and cor nd corruption. ents and margi ites are include	ruption are ar We make use nal effects at 1 ed. For detaile	e taken into con of a Probit mo neans. Sample ed variable deso	asideration. Wy odel with robus period is 20000 cription see Ta	e run the regres st standard err 11-2015q4. All ble 1. The *, *	sions separately ors clustered by financial control * and *** signs
denote statistical significance at	the $1,5$ and $10$	1% level.						

Table 16: State economic freedom and corruption

	()		(2	(7	(3			(4)
	Religi	osity	Religi	iosity	Political po	olarization	Re	ligiosity
	(Above	average)	(Below <i>i</i>	average)	(Above	$\mathbf{verage}$	(Belor	w average)
	$\beta / SE$	Mfx	$eta \ / \ { m SE}$	Mfx	$\beta / SE$	Mfx	$\beta$ / SE	Mfx
Committee Chair	-0.106	-0.001	-0.213***	-0.002***	-0.290***	-0.003***	-0.052	-0.000
	(700.0)		(0.073)		(0.078)		(0.093)	
Risk-weighted assets ratio	-0.173	-0.001	0.018	0.000	-0.122	-0.001	0.064	0.001
	(0.188)	-	(0.121)		(0.125)		(0.171)	
Risk-based capital ratio	-2.777***	$-0.019^{***}$	-3.756***	$-0.040^{***}$	$-4.135^{***}$	-0.038***	$-2.440^{***}$	$-0.020^{***}$
Dotum on accord	(0.475)	0 171 ×**	(0.446)	***000 U	(0.400)	0 1 00 ***	(0.501)	0 166***
CODEED TID TIMODI	(0.996)	101.0-	(0.740)	607.0-	(106.01)	POT-0-	(0.882)	001.0-
Liquidity	-0.114	-0.001	$1.308^{***}$	$0.014^{***}$	$1.131^{***}$	$0.010^{***}$	0.347	0.003
	(0.368)		(0.201)		(0.213)		(0.303)	
Size	0.028	0.000	$0.034^{***}$	$0.000^{***}$	$0.043^{***}$	$0.000^{***}$	0.023	0.000
	(0.021)		(0.011)		(0.013)		(0.017)	
GDP	$0.194^{*}$	$0.001^{*}$	$0.550^{***}$	$0.006^{***}$	$0.594^{***}$	$0.006^{***}$	$0.234^{**}$	$0.002^{**}$
	(0.090)		(0.091)		(0.103)		(0.100)	
Population	-0.304**	-0.002**	-0.498***	-0.005***	-0.624***	-0.006***	-0.183	-0.001
	(0.136)		(0.106)		(0.127)		(0.120)	
Constant	-0.144		-1.455*		0.038		$-2.401^{***}$	
	(1.200)		(0.827)		(0.998)		(0.875)	
Observations	162427		197777		197630		163370	
Cluster	3866.000		5156.000		6030.000		5007.000	
Pseudo $R^2$	0.111		0.123		0.133		0.106	
Time dummies	$\mathbf{YES}$		$\mathbf{YES}$		YES		YES	
State dummies	$\mathbf{YES}$		$\mathbf{YES}$		$\mathbf{YES}$		YES	
Table 17 reports the estimation sub-samples of states with high, all model specifications we repo one quarter. A full set of time d	results when st low religiosity ort estimated co lummies are inc	ate religiosity a and political p oefficients and cluded. For det	and political polarization. W olarization. W marginal effect cailed variable	olarization are é make use of ts at means. S description see	are taken into a Probit model bample period : Pable 1. The	consideration. with robust si is 2000q1-2015 *, ** and ***	We run the regr tandard errors cl iq4. All financial signs denote stat	essions separately for ustered by bank. For l'variables are lagged istical significance at
the $1,5$ and $10\%$ level.								

Table 17: State religiosity and political polarization

# Appendix I

## Enforcement actions classification

ENFORCEMENT ACTIONS ISSUED A	GAINST BANKS
A.1. SEVERE ( ordered from most seven	re to less severe):
Deposit Insurance Termination /Threat	Decision to threat to suspend or terminate a bank's deposit insurance scheme by the FDIC, when unsound and unsafe banking practices are detected or when violations of laws and regulations have taken place. Deposit Insurance Termination can be imposed if a bank has neglected previous enforcement actions issued against the bank.
Cease and Desist Order	Banks that receive Cease and Desist orders are required to follow specific actions outlined by their primary supervisor. C&D orders can be enforced by law, in the federal banking system. Typical reasons or the issuance of C&D orders are the engagement in unsafe and unsound activities, violations of laws and regulations. A C&D may impose specific orders to stop the bank engaging in specific banking practices or may outline a particular strategy in order to improve asset quality, promote growth, decrease risk, etc.
Formal written agreement	The institutions subject to this type of action, enter into an agreement with their primary regulator to take particular actions or to follow particular proscriptions in written agreement. Unlike the C&D orders, although FAs are also legally enforceable, they are however, not enforceable through the federal court. FAs can nonetheless lead to the issuance of Civil Money Penalties, when they are ignored. Reasons that FAs are imposed are unsound practises, mismanagement policies, or "insider" abuse. FAs can lead to more severe types of enforcement actions if not taken into consideration.
Prompt Corrective Action	Prompt Corrective Actions are issued usually when undercapitalization issues are detected. These actions order banks on taking remedial actions in order to overcome the deficiencies in their level of capital. Among the corrective measures outlined, in some cases there may be dismissal of management, restrictions on executive payments, asset growth, rates paid on deposits or even prohibition on certain activities, such as approval for acquisition deals from the regulatory authorities.
A.2. LESS SEVERE TYPE	
Civil Money Penalty (CMP)	Monetary penalties against banking institutions that engage in unsafe or unsound banking practices, violations of laws or failure to comply with an order issued previously.
Call report penalty (CR-P)	Monetary penalties against banking institutions that fail to file Call Reports on time or in accordance to the general outline or even for misreporting information on Call Report files.